



#### International Journal of Banking, Accounting and Finance

ISSN online: 1755-3849 - ISSN print: 1755-3830 https://www.inderscience.com/ijbaaf

## Hedge fund activism and short-term value creation prior to the initial filing: evidence from US firms

Efstathios Karpouzis, Dimitris Margaritis, Maria Psillaki, Christos Staikouras

DOI: 10.1504/IJBAAF.2022.10051472

#### **Article History:**

Received:	
Accepted:	
Published online:	

16 September 2021 11 September 2022 06 March 2023

# Hedge fund activism and short-term value creation prior to the initial filing: evidence from US firms

### Efstathios Karpouzis\*

Department of Management Science and Technology, University of Peloponnese, Sehi, Tripoli, 22100, Greece Email: karpouzis@go.uop.gr \*Corresponding author

## **Dimitris Margaritis**

Department of Accounting and Finance, The University of Auckland, 12 Grafton RD, Auckland 1010, New Zealand Email: d.margaritis@auckland.ac.nz

## Maria Psillaki

Department of Economics, University of Piraeus, 80, M. Karaoli and A. Dimitriou St., Piraeus 18534, Greece and Department of Economics and Business, Neapolis University Paphos, 2 Danais str. 8042, Paphos, Cyprus Email: psillaki@unipi.gr

## Christos Staikouras

Department of Accounting and Finance, Athens University of Economics and Business, 76 Patission St., Athens 10434, Greece Email: cstaik@aueb.gr

**Abstract:** Hedge funds trade on non-public information which is not clearly disclosed to the market at the time of schedule 13D or 13G filings. Using a hand-collected dataset of US hedge funds interventions we provide new evidence on the value of non-public information at the time the filer surpasses the 5% threshold and the filing obligation is triggered. We find returns are abnormally high prior to the disclosure and more importantly prior to the 5% threshold date but only for schedule 13D events. Both 13D and 13G targeted firms generate abnormal returns in the post filing period which are significantly lower than the 13D returns prior to the filing date. The novelty of our approach is that it distinguishes gains resulting from insider trading around the 5% threshold event from those generated by information asymmetry around the filing event.

#### *E. Karpouzis et al.*

**Keywords:** hedge fund activism; asymmetric information; insider trading; abnormal returns; event studies.

JEL codes: G12, G23, G34.

**Reference** to this paper should be made as follows: Karpouzis, E., Margaritis, D., Psillaki, M. and Staikouras, C. (2023) 'Hedge fund activism and short-term value creation prior to the initial filing: evidence from US firms', *Int. J. Banking, Accounting and Finance*, Vol. 13, No. 3, pp.277–320.

**Biographical notes:** Efstathios Karpouzis holds a PhD in Finance from the Department of Economics at the University of Piraeus. He is currently an Adjunct Lecturer at the University of Peloponnese, Department of Management Science and Technology. His research interests focus on finance and especially on alternative forms of financing.

Dimitris Margaritis is a Professor of Finance at the University of Auckland Business School. He served as an advisor, the Head of Research and Senior Research Fellow at the Reserve Bank of New Zealand and was a member of the World Bank Project on Financial Reform. He previously taught at the University of British Columbia, University of Washington, Southern Illinois University, State University of New York at Buffalo, Waikato Management School and AUT University. His research portfolio covers diverse areas including several contributions to the literature on firm efficiency and productivity, money and banking, corporate finance and asset pricing. He has published two books and more than 70 journal articles mainly in A\*/A (ABDC ranked) journals.

Maria Psillaki is a Professor of Economics and Finance at the Department of Economics of the University of Piraeus and Senior Research Fellow at Neapolis University Pafos, (Cyprus). She has held visiting research positions in London (Birckbeck College, University of London) as a visiting researcher, funded by the (CEPR) and the (ERSC) and in Chicago as a visitor researcher at the Graduate Booth School of Business, University of Chicago. She has held visiting faculty positions at European and American universities. She has published in high quality peer reviewed journals and she served as a referee in a number of journals.

Christos Staikouras is a member of the Greek Parliament since 2007, appointed as the Minister of Finance, since 2019. He is an Associate Professor in Finance at the Athens University of Economics and Business. He has published many research papers in international refereed scientific journals, books and monographs and he has also participated in many scientific international conferences. His principal areas of research include banking and finance.

#### 1 Introduction

Is chasing a 13D or 13G targeted stock immediately after the filing the best course of action for investors? Financial markets have always been driven by differences in how much people know. Better informed investors have an advantage, for example, as a result of superior skill or by acquiring pertinent information before others, which they may be able to exploit (see Hirshleifer et al., 1994). As Grossman and Stiglitz (1980) put it 'those

who expend resources to obtain information do receive compensation'. Such information asymmetries do present opportunities to investors for earning abnormal returns. Hedge fund activism introduces an asymmetry between insiders and outside investors and as shown by prior literature hedge funds trade based on private information (Massoud et al., 2011; Ivashina and Sun, 2011; Agarwal et al., 2013).

In this study we exploit Securities and Exchange Commission (SEC) disclosure requirements to examine whether hedge funds hold valuable non-public information resulting from their internal assessments of active and passive targets around disclosure and pre-disclosure dates. Hence we go beyond established evidence on the value of asymmetric information around the filing date by focusing on abnormal returns at the earlier time when the fund actually intervenes in the market.<sup>1</sup> We carry out the analysis using a hand-collected dataset of Schedule 13D ('active') and Schedule 13G ('passive') filings made by the same group of hedge funds during the 2005–2013 period. Capturing both active and passive block holdings by the same hedge fund is an important feature of the research design since it helps identify more clearly the effects of activism on short-term value creation (see Clifford, 2008).

We focus on hedge funds as shareholder activists since they are more 'eclectic investors' (Klein and Zur, 2009), experienced and sophisticated (Lipton, 2013), managed by highly incentivised managers (Clifford, 2008), and are subject to less SEC regulations than other funds. More precisely, they are not required to register with the SEC or make periodic reports like mutual funds or pension funds. Hedge funds follow high risk investment strategies including the use of leverage, 'short-selling' or other speculative investment practices. Furthermore, they set a minimum level of income or assets for investors and charge high fees. These characteristics, their organisational form and managerial incentives make hedge funds a considerable threat for target firms and ultimately the market (Clifford, 2008).

The Schedule 13D filing date has attracted significant attention in prior studies since investors have to disclose their ownership stake with the SEC revealing information about the purpose of transaction and their identity. Schedule 13D filings are often seen by investors as a signal that the targeted stock is undervalued and poised to appreciate. The vast majority of the previous literature provides evidence of short-term value creation (e.g., Becht et al., 2008, 2017; Brav et al., 2008; Clifford, 2008; Klein and Zur, 2009; Greenwood and Schor, 2009; Boyson and Mooradian, 2011; Gow et al., 2014; Bebchuk et al., 2013; Collin-Dufresne and Fos, 2015) and a run-up in the stock return before the initial filing (Brav et al., 2008; Collin-Dufresne and Fos, 2015). For example, Brav et al. (2008) find that price and abnormal trading volume increase about ten days prior to the 13D filing date while Collin-Dufresne and Fos (2015) report a 6% cumulative abnormal return in the (t - 10, t + 1) window around the filing date and about 3% in the (t - 1, t + 1) window.

Schedule 13G filings have received less attention in the literature. Yet acquiring 5% or more of a stock, a 13G investor may be signalling that a stock is a good buy and it is likely to appreciate in value. Giglia (2016) notes that investors could also 'hide' their 'active' purposes through a Schedule 13G filing for short-term gains. She points out the inappropriate use of Schedule 13G filings by investors who have 'active' purposes, silently increasing their stake at lower cost in order to avoid market and regulatory scrutiny. However, if 13G filers follow this strategy, we should notice similar

abnormalities to 13D filers in returns, volume and volatility during the stock building phase, especially around the *event* date that the filing obligation is triggered.

Gantchev and Jotikasthira (2018) used activist hedge funds' 13F filings to estimate activist toeholds in potential targets, highlighting the possibility of widespread activism threat propagation beyond the firms being targeted. However, a 13F holding does not necessarily lead to an active or a passive intervention in the near future. Beyond that, hedge funds are generally reluctant to reveal their private information on target firms during the stock building period in order to avoid 'free-riding' from other market participants. There is also a significant number of Schedule 13D filings where hedge funds exit the target company without stating their plans or proposals or even changing their beneficial ownership. And the target stock may have advanced significantly during the days prior to the filing leaving unsuspected investors with plunging prices. Clifford (2008) examines the effects of 'active' versus 'passive' activism by hedge funds and finds that hedge funds with 'active' blocks earn larger excess stock returns and improve performance more than a control group of the same hedge funds with 'passive' blocks.

In this paper we examine whether hedge funds hold valuable non-public information for their active and passive targets. We analyse whether hedge funds' non-public information is reflected into the target firm's abnormal returns, returns volatility and volume prior to an event date and explore whether the filer's intentions matter. What sets apart this study from much of the previous literature is that we specifically assess the importance of insider information at a specific event date prior to the filing date with the SEC. The event date is the time the hedge fund surpasses the 5% threshold and a filing event is triggered. Hence our approach provides a more powerful test of the value of a hedge fund's internal assessment of target firms. We carry out our empirical tests using information on both 13D and 13G filings made by the same group of hedge funds. More specifically, we test the following hypotheses for both 13D and 13G filings at the trigger event and filing dates:

- H1 The stock price abnormal return to hedge fund interventions is expected to be zero on the trigger event (filing) date.
- H2 The stock price abnormal volume to hedge fund interventions is expected to be zero on the trigger event (filing) date.
- H3 The stock price abnormal volatility to hedge fund interventions is expected to be zero on the trigger event (filing) date.

We exclude from our sample events where the filing date matches or is very close (one day) from the event date in order to avoid possible abnormalities caused by the filing revelation. We calculate abnormal returns using standard event study methodology. Furthermore, we calculate abnormal return volatility (AVAR) and abnormal volume for Schedule 13D filings, Schedule 13G filings and for control samples of similar non-targeted firms based on industry (three-digit SIC), exchange, and size. The results show the existence of statistically significant and positive abnormal returns, AVAR and volume at the (-10, 0), (-5, 0) and (-1, 0) windows prior to the event date, and for the (-45, 0) window prior to the filing date, albeit abnormal returns are significant only for Schedule 13D filings. Our findings suggest that hedge funds have non-public information on their 'active' targets during the stock building window prior to the official disclosure with the SEC. Abnormal returns are positive and statistically significant after the filing

date for both 13D and 13G target firms. However, these returns are significantly lower compared to the 13D returns prior to the filing date.

Our findings are consistent with recently proposed changes by the SEC<sup>2</sup> for amendments to the rules governing beneficial ownership reporting under Exchange Act Sections 13(d) and 13(g). There is no clear indication in our data whether investors intervened in a firm actively through 13D filings, since they exited without stating specific plans or proposals. Although as a beneficial owner, activists might have had conversations with the management, in our case this kind of information was never made publicly available through official filings. On the other hand, while the 13G filings were intended for 'passive investors', in our setting they appear very much alike the 'active' 13D filings in terms of publicly available information. With no sightings of 'active' plans or proposals in the 13D filings, our findings of much higher abnormal returns during the period leading to the Schedule 13D filing date raise concerns about the effects of information asymmetries on market efficiency.<sup>3</sup>

The rest of the paper is organised as follows. Section 2 describes the institutional background and construction of the data sample. Section 3 describes the methodology, and Section 4 reports the results. Section 5 concludes.

#### 2 Institutional background and data description

Under Rule 13d-1(a) of the 1934 Securities Exchange Act, investors who hold more than 5% of any class of securities of a publicly traded company and intend to influence the control of a target company must file a Schedule 13D form, which must be submitted to the SEC within 10 days. However, under Rule 13d-1(c), SEC provides the 'passive investor' exemption under which investors who

- a have no intention to intervene in the company's internals
- b are not qualified institutional investors
- c hold less than 20% of a company's common stock, may file a short-form statement on Schedule 13G.<sup>4</sup>

We construct a hand-collected data set of US hedge funds activist interventions over the period 2005–2013, which records both the Schedule 13D and Schedule 13G filing dates. We use the Historical SEC Edgar Archives from the Edgar Database of the SEC (https://www.sec.gov/cgi-bin/srch-edgar) and collect the Schedule 13D filings first. We then limit the filers to hedge funds only and collect the Schedule 13G filings made by these hedge funds.

Previous literature uses a variety of processes to identify hedge funds. In this paper, we form our sample of hedge funds following Brunnermeier and Nagel (2004) and Griffin and Xu (2009). The procedure we follow, while more laborious, is an important part of the research design since it helps us identify 'pure' hedge funds in the sense of Brunnermeier and Nagel (2004) and Griffin and Xu (2009). We search the Investment Adviser Public Disclosure website (https://www.adviserinfo.sec.gov/IAPD/default.aspx) for each of the 'Reporting Persons'<sup>5</sup> of the Schedule 13D filings and include in our sample only firms that were registered as investment advisers with the SEC and thus,

filed an ADV form.<sup>6</sup> The ADV form must have been filed by investment advisers who manage more than \$25 million in assets.<sup>7</sup>

We then search the 'Item 5' section of the ADV form. In this section investment advisers are required to disclose information about their business, employees, clients, fee schedule, disciplinary events involving the advisers and their employees, and conflicts of interest. We focus on the 'clients' subsection, where the advisers disclose information about the approximate number of their clients along with the amount of total regulatory assets under management.<sup>8</sup>

The next step is to examine the 'compensation arrangements' subsection where filers report their compensation types, i.e.

- a percentage of assets under management
- b hourly charges
- c subscription fees (for a newsletter or periodical)
- d fixed fees (other than subscription fees)
- e commissions
- f performance-based fees
- g other.

We include in our sample only those firms that at least 50% of their clients were either 'pooled investment vehicles (other than investment companies and business development companies)' or 'high net worth individuals' or a combination of the above and charged performance-based fees.

We kept in our sample Schedule 13G filings under Rule 13d-1(c) but we excluded financial firms<sup>9</sup> owing to differences in the information flow. Clifford (2008) includes in his sample Schedule 13D filings with specific plans and proposals which offers investors additional information about the firm. We differentiate from Clifford (2008) by limiting our sample to the subsample of Schedule 13D filings which did not state specific plans or proposals because the information on the target firm is twofold: reflecting the new ownership and the hedge fund's plans, simultaneously. Under this limitation, in both filing type's hedge funds hold more than 5% on the target firm and only the investors' future plans on the target firm differ. Under this caveat, Schedule 13D filers have three possible tactics to follow after the initial filing:

- a hold or increase/decrease their shares without stating specific plans or proposals
- b file an amendment stating specific plans or proposals
- c exit.

On the other hand, Schedule 13G filers may:

- a hold or increase/decrease their shares
- b switch to 'active' filing a Schedule 13D with or without specific plans and proposals
- c exit.

So, in both cases hedge funds have a complete discretion on when and whether to follow a more aggressive agenda by simply filing an amendment (13D/A) or switch filing (13G

to 13D). Furthermore, our dataset does not contain 'wolf pack' Schedule 13D filings and Schedule13G filings converted to Schedule 13D filings. We also exclude Schedule 13D filings where ownership of more than 20% was stated. We identified 151 'pure-play' hedge funds (Ben-David et al., 2013), which filed a total of 640 Schedule 13D filings and 1011 Schedule 13G filings.

We match targeted firms with a control sample of similar firms based on year, exchange, industry (three-digit SIC) and size (market value). In particular, we use the coarsened exact matching restricting control firms to the same stock exchange, three-digit SIC and size. Then, we identify the nearest neighbour of each targeted firm. We exclude from our sample events that did not have a match. If the nearest neighbour had missing values, we moved to the next one.

Finally, we perform two robustness tests. First, we allow for possible seasonality effects using the same calendar windows for the target firms one year earlier. Second, we match 13D filings with a control sample of 13G filings using different matching criteria. In particular, we follow Brav et al. (2018) and create a matching sample based on year, industry (two-digit SIC), (log) market value, market-to-book ratio, return on assets (ROA) measured at t - 1, and the change in the target firm's ROA measured between years t-3 and t-1.

The stock returns were downloaded from the CRSP database and the exchange, industry and fundamentals are from Compustat.

#### 3 Methodology

We examine the market's reaction to hedge fund activist interventions using an eventstudy methodology. We assume that capital markets are efficient and examine the impact of new information on the stock's return. We compare Schedule 13D with Schedule 13G filings in the (-45, 0) window where t = 0 is the filing date. We define the date where the hedge fund surpasses the 5% threshold as the event date. Investors have a ten-calendar day<sup>10</sup> period to disclose their new holdings with the SEC either with a Schedule 13D or with a Schedule 13G filing.



We identify as new information the possible private information through a non-publicly available channel (event date) and the publicly available information from the SEC (filing date). Consistent with the hypothesis that hedge fund manager protect proprietary information (Aragon et al., 2013), we assume that hedge funds do not leak their beneficial ownership prior to the event date. We explore whether there exist statistically significant abnormal returns, return volatility and volume for the targeted firm between 'active' and 'passive' interventions in terms of filing type (Schedule 13D vs. Schedule 13G). Due to their nature, Schedule 13D filings contain more information about the transactions. In order to avoid bias in discriminating between 13D and 13G abnormal returns, we limit the sample of Schedule 13D filings to those where filers do not disclose any plans or proposals about the target firm. These filings are comparable with the

Schedule 13G filings where the assumption of 'not influencing the firm' stands due to the filing's nature. In this sense, we are able to test any differences that may exist in the reaction of the market to the 13D and 13G disclosures on a level playing field. The filing event period covers 45 pre-filing trading days<sup>11</sup> and the filing date (-45, 0). We get 45 trading days prior to the filing date since 13D filers are obliged to report in detail their transactions in the item 5c section. However, this information is not available for 13G filings. We exclude from our sample events where the filing date matches or is within one day from the event date to avoid possible abnormalities caused by the filing revelation.

Abnormal returns are assessed in terms of realised returns and the returns that would normally be expected by the market. We follow Brown and Warner (1980), who define an abnormal return  $AR_{it}$  as the difference between the actual return of stock *i* at the event day *t* and the expected stock return at the event day *t* predicted by an estimated asset pricing model:

$$AR_{it} = R_{it} - E(R_{it}) \tag{1}$$

where  $R_{it}$  is the actual return of stock i at event day t, i = 1, 2, ..., N, with N denoting the total number of stocks, while  $E(R_{it})$  denotes the expected stock return at time t. The expected returns represent the returns that would have been observed during the event window in the absence of the event. We set the estimation window at a fixed length of 206 actual trading days, comprising 252 to 46 trading days prior to the filing date (where t = 0 is the filing date).

We calculate expected returns using the market model, the Carhart four-factor model, and the Fama-French five-factor (FF5) model. The market model is represented by the following linear regression model estimated by least squares:

$$R_{it} = \alpha_0 + \beta_1 R_{mt} + u_{it} \tag{2}$$

where  $R_{mt}$  is the return on the market, defined as the value-weighted return of all CRSP firms incorporated in the USA and listed on the NYSE, AMEX, or NASDAQ that have a CRSP share code of 10 or 11 at the beginning of month *t*, share information and price data also at the beginning of month *t*.

The Carhart and FF5 models are extensions of the Fama-French three-factor (FF3) model. The Carhart model adds momentum (MOM) to the three Fama-French factors: market ( $R_{mt} - R_{fi}$ ), size (SMB) and value (HML). The FF5 model adds robust minus weak (RMW) profitability and conservative minus aggressive (CMA) investment to FF3. The Carhart and FF5 models are given, respectively, by:

$$R_{it} - R_{ft} = \alpha_0 + \beta_1 \left( R_{mt} - R_{ft} \right) + \beta_2 SMB_t + \beta_3 HML_t + \beta_4 MOM_t + u_{it}$$
(3a)

$$R_{it} - R_{ft} = \alpha_0 + \beta_1 \left( R_{mt} - R_{ft} \right) + \beta_2 SMB_t + \beta_3 HML_t + \beta_4 RMW_t + \beta_5 CMA_t + u_{it}$$
(3b)

where  $R_{ft}$  is the one-month T-bill rate.

We calculate standardised abnormal returns ( $SAR_{it}$ ) following Patell (1976) and Dodd and Warner (1983) by normalising each firm's abnormal return by the square-root of its estimated forecast variance:

$$SAR_{it} = AR_{it} / s_{AR_{it}}$$
<sup>(4)</sup>

where

$$s_{AR_{it}} = \left\{ s_i^2 \left( 1 + \frac{1}{L_i} + \frac{(R_{mt} - \overline{R})}{\sum_{t=1}^{L_i} ((R_{mt} - \overline{R})^2)} \right) \right\}^{1/2}$$
(5)

Here  $s_i^2$  is the estimated residual variance from the market, Carhart or FF5 model for stock *i*, and  $\overline{R}$  is the average market return over the  $L_i$  days used for the regression. The standardised cumulative abnormal returns over the event period are obtained by the summation of the standardised abnormal returns over the event window divided by the square root of the number of days in the window. The resulting standardised cumulative excess return (SCAR) is assumed to have a standard normal distribution with zero mean and unit variance. To test the significance of the average standardised cumulative excess return (SCAAR) in a sample of *N* stocks we use a standard normal statistic:

$$Z = SCAAR\sqrt{N} \sim N(0,1) \tag{5}$$

In order to calculate short-term AVAR, we follow Landsman and Maydew (2002), DeFond et al. (2007), Landsman et al. (2012) and Devos et al. (2015) who define AVAR  $(AVAR_{it})$  as the stock return variance  $(\overline{AR_{it}^2})$  over the event window, scaled by the stock return variance  $(s_i^2)$  over the estimation window. This is calculated as:

$$AVAR_{it} = \overline{AR_{it}^2} / s_{it}^2 \tag{6}$$

The stock return variance over the event window is the mean squared abnormal return of each firm over the examined event windows divided by the variance of each firm's residuals calculated over the (-252, -46) estimation period. When AVAR is greater than 1, the firm's stock has greater than normal volatility.

We estimate mean-adjusted abnormal trading volume changes associated with activist interventions following Ajinkya and Jain (1989), Cready and Ramanan (1991) and Campbell and Wasley (1996) who define mean-adjusted abnormal trading volume as the share turnover rate minus the normal turnover rate measured over the estimation window, -252 trading days to -46 days relative to the filing date:

$$AV_{it} = V_{it} - \overline{V_i}$$

where  $V_{it}$  is the number of shares of firm *i* traded on day *t* divided by number of shares outstanding on the event window and  $\overline{V}_i$  is the mean trading volume over the estimation period.

#### 4 Empirical results

Table 1 reports descriptive statistics of firm specific and filer specific information. Hedge funds tend to target companies that are generally 'value' firms, with low market value relative to book value. Table 1 show that more activist 13D filers target on average firms with lower market to book ratios than 13G filings. At the event date, 13D target firms have on average larger market values of than 13G firms. Schedule 13D target firms have

higher ROA with a median value of 3.94% compared to 2.85% for 13G firms. Also, at the event date, 13D and 13F filers own 6.58% and 7.47%, respectively, of the shares outstanding. The filing delay, which is the gap between the event and the filing date, for the full sample (clean events) is 8.50 (6.91) days for Schedule 13D filings and 13.82 (7.48) for Schedule 13G filings.

SC	C 13D (640 e	events)			
	Mean	Std. dev.	p.25	p.50	p.75
Firm characteristics					
MV (in millions)	1,244.65	2,044.25	154.28	432.64	1,423.93
ROA	-5.80	70.20	-3.72	3.94	8.29
Market-to-book ratio	0.39	0.58	0.08	0.16	0.44
Event characteristics					
Initial stake (in percent)	6.58	2.09	5.25	5.71	7.11
Filing delay – full sample (days)	8.5	4.74	6	9	10
Filing delay – clean events (days)	6.91	3.21	4	8	10
SC	13G (1011	events)			
	Mean	Std. dev.	p.25	p.50	p.75
Firm characteristics					
MV (in millions)	972.34	1,506.97	135.30	400.08	1,121.69
ROA	-8.89	41.45	-11.70	2.85	8.26
Market-to-book ratio	0.79	0.98	0.24	0.42	0.89
Event characteristics					
Initial stake (in percent)	7.47	2.31	5.98	6.58	8.12
Filing delay – full sample (days)	13.82	19.78	7	10	11
Filing delay – clean events (days)	7.48	2.74	6	8	10

Table 1Descriptive statistics

#### 4.1 Filing date

Tables 2–4 report the cumulative average abnormal returns (CAARs), standardised cumulative average abnormal returns (SCAARs), AVAR and cumulative mean-adjusted abnormal volume (CAAV) for the 45-trading day window prior and after the filing date, and the 91-trading day window surrounding the filing date. We compare the Schedule 13D with Schedule 13G events (Table 2), Schedule 13D events with a control sample of firms (Table 3), and Schedule 13G events with a control sample of firms (Table 3), and Schedule 13G events with a control sample of firms (Table 4). The control sample is based on year, exchange, industry (three-digit SIC) and size. First, we examine the full sample. Then we exclude events where the ten-calendar day window is violated, and create a subsample of 'clean' events. We also exclude from our sample the events where the filing date coincides with the event date.

Table 2 presents CAARs, SCAARs, AVAR and CAAV for the 91-trading day window surrounding the filing date as well as for 45-trading days prior and after the filing date for Schedule 13D and Schedule 13G events. The main difference of these filings is that Schedule 13G filers have no intention of influencing control over the stock

issuer. However, prior to the initial filing, market participants are not aware of the forthcoming disclosure filing and/or filing type, so we expect Schedule 13D and Schedule 13G events to perform similarly.

In Panel A we present the results for the full sample. In Panel B we exclude events where the 10-calendar day disclosure period is violated to avoid biasing the computed returns and volume statistics. CAARs, SCAARs, AVAR and CAAV are positive and statistically significant for Schedule 13D filing during the pre-and post-filing periods. In particular, the CAAR for Schedule 13D filings using the market model, as shown in column (1), is 12.4% for the (-45, 0) pre-filing window and is statistically significant at the 1% level and much higher than the 4.3% cumulative abnormal return in the post-filing (0, +45) period. Schedule 13G filings have slightly positive CAARs of 1.2% and 3.2% in the pre and post filing periods, respectively, albeit only the latter CAAR is statistically significant. Interestingly, Schedule 13D abnormal returns are much higher than Schedule 13G returns for all the examined measures with the difference (Schedule 13D minus Schedule 13G) being positive and statistically significant at the 1% level during the pre-filing period, and for the 91-trading day window surrounding the filing date. Notably, the proportion of the positive Schedule 13D abnormal returns is 66.71% (67.15%) for the full sample ('clean' events) which in both cases are higher when compared with the Schedule 13G filings. We obtain similar results using the Carhart and Fama-French five-factor models as shown in columns (2)-(3), respectively, and when we restrict our sample to 'clean' events. AVAR and volume for 13D filings are larger compared to 13G filings but the differences are significant only in the pre-filing window.



Figure 1 CAARs surrounding the initial filing – full sample (see online version for colours)

Note: The y axis measures the CAARs in percent for schedule 13D and Schedule 13G filings calculated using the market model for the period (-45, 45) where t = 0 is the filing date.

			Schedule 13D	Panel A: F	iling date – All ev	ents Schedule 13G			Difference	
	t = Filing date	(t)	(2)	(3)	(t)	(2)	(3)	(t)	(2)	(3)
CAAR	(-45, 0)	0.124***	0.120***	0.120***	0.012	0.010	0.008	0.112***	$0.110^{***}$	0.110***
	Pos: Neg	427:213	423:217	418:222	486:525	477:534	479:536			
	(0, +45)	$0.043^{***}$	$0.04I^{***}$	0.039***	0.032***	$0.034^{***}$	0.035***	0.011	0.006	0.004
	Pos: Neg	359:281	364:276	353:287	544:467	556:455	563:462			
	(-45, +45)	0.158***	0.153***	0.152***	0.042***	0.042***	$0.04I^{***}$	0.117***	0.112***	0.111***
	Pos: Neg	426:214	431:209	420:220	539:472	542:469	532:479			
SCAAR	(-45, 0)	0.368***	$0.374^{***}$	$0.380^{***}$	-0.007	-0.009	-0.011	0.375***	$0.384^{***}$	0.039***
	(0, +45)	$0.114^{***}$	0.115***	0.112***	0.077***	$0.087^{***}$	$0.087^{***}$	0.038	0.029	0.025
	(-45, +45)	0.458***	0.465***	0.468***	0.065***	$0.071^{**}$	0.069***	0.393***	$0.394^{***}$	0.399***
AVAR	(-45, 0)	3.402***	3.653***	3.743***	1.867***	2.007***	2.069***	I.535***	1.645***	1.674***
	(0, +45)	1.375***	1.493***	1.536***	1.539***	<i>I.657</i> ***	$1.684^{***}$	-0.163	-0.164	-0.148
	(-45, +45)	2.389***	2.574***	2.640***	1.708***	<i>I.839</i> ***	1.883***	0.681***	0.735***	0.757***
CAAV	(-45, 0)	$0.308^{***}$			0.163***			0.145***		
	(0, +45)	$0.118^{***}$			$0.070^{***}$			0.049*		
	(-45, +45)	$0.418^{***}$			0.229***			0.188***		
Number of	obs.	640	640	640	1,011	1,011	1,011			
Notes: The t using the re	able presents CAARs the market model (1) sults for the full samp	, SCAARs, AVA ), the Carhart moc ple. Panel B prese	R, and CAAV f. del (2), and the F ants the results fc	or Schedule 13D, <sup>2</sup> F5 model (3). In or the 'clean' ever	Schedule 13G fil each case, we cal ats excluding the e	ings, and their c culate the numb events where th	lifference, 45 day er of positive an e ten-day disclos	ys before and afte d negative events ure period is viol	er the filing dat s. Panel A pres lated. *, **, ***	e ents *
indic	ate statistical significs	ance at the $10\%$ , $10\%$	5% and 1% level	l, respectively.						

## Table 2Abnormal returns, return volatility and volume of Schedule 13D filings, Schedule<br/>13G filings and their difference

(2) 0 124***	(3)	(l)	(2)	(3)	(1)	(7)	(2)
0 124***						~ ~	
	0.123***	0.008	0.006	0.005	$0.118^{***}$	$0.117^{***}$	$0.118^{***}$
322:159	318:163	320:316	313:353	313:353			
0.031***	$0.029^{***}$	0.039***	$0.043^{***}$	$0.044^{***}$	-0.004	-0.012	-0.015
262:219	253:228	360:306	369:297	373:301			
0.146***	0.143***	$0.043^{**}$	$0.045^{***}$	0.045**	$0.108^{***}$	$0.101^{***}$	$0.098^{***}$
318:163	309:172	360:306	362:304	353:311			
0.382***	$0.388^{***}$	-0.007	0.010	-0.010	$0.380^{***}$	0.392***	0.398***
0.087***	$0.083^{***}$	0.086***	$0.104^{***}$	0.105***	0.003	-0.017	0.100
$0.44I^{***}$	0.442***	0.072*	0.085***	0.085***	0.362***	0.356***	0.357***
3.577***	3.668***	<i>I.722</i> ***	1.820***	<i>I.857</i> ***	$I.600^{***}$	1.757***	1.831***
1.146***	1.181***	1.666***	1.775***	1.805***	-0.608*	-0.629*	-0.624*
$2.360^{***}$	2.423***	1.698***	1.803***	1.836***	0.490 * *	0.558**	$0.587^{**}$
		0.203***			0.121***		
		0.106***			0.000		
		0.304***			0.116**		
481	481	666	666	999			
	0.140 0.382*** 0.087*** 0.441*** 3.577*** 2.360*** 2.360***	0.1+0 0.1.1+0 0.1.1+3 0.1.1+3 0.1.1+2 0.382*** 0.382*** 0.087*** 0.083*** 0.442*** 3.568*** 1.141+** 0.442*** 3.568*** 2.360*** 2.423*** 2.423*** 2.423*** 2.423***	0.203 *** 0.423*** 0.087 *** 0.087 *** 0.083 *** 0.087 *** 0.087 *** 0.087 *** 0.086 *** 0.072 *** 0.442 *** 0.072 *** 0.442 *** 0.070 *** 0.070 ***	0.140 $0.145$ $0.045$ $0.045$ $318:163$ $309:172$ $360:306$ $362:304$ $0.382***$ $0.388***$ $-0.007$ $0.010$ $0.382***$ $0.388***$ $0.007$ $0.010$ $0.387***$ $0.388***$ $0.007$ $0.010$ $0.377***$ $0.388***$ $0.086***$ $0.104***$ $0.442**$ $0.072*$ $0.003$ $0.86***$ $3.577***$ $3.668***$ $1.722***$ $1.829***$ $2.360***$ $2.423***$ $1.666***$ $1.775***$ $2.360***$ $2.423***$ $1.668***$ $1.803***$ $0.106***$ $0.106***$ $0.304***$ $0.304***$	0.140       0.142       0.042       0.044       0.044         318:163       309:172       360:306       362:304       353:311         0.382***       0.383***       -0.007       0.010       -0.010         0.87***       0.383***       0.086***       0.104***       0.001         0.87***       0.083***       0.085***       0.005       362:304       353:311         0.87***       0.085***       0.086***       0.104***       0.005       0.010         1.41***       0.442***       0.072*       0.085***       0.085***       0.085***       0.065***         3.577***       3.668***       1.722***       1.820***       1.855***       2.855***         1.441***       1.181**       1.666***       1.803***       1.805****       2.805****         2.360***       2.423***       1.668***       1.803***       1.836***       1.836***         0.106***       0.106***       0.303***       1.836***       0.866       666 </td <td><math display="block">\begin{array}{cccccccccccccccccccccccccccccccccccc</math></td> <td><math display="block">\begin{array}{cccccccccccccccccccccccccccccccccccc</math></td>	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$

Table 2Abnormal returns, return volatility and volume of Schedule 13D filings, Schedule<br/>13G filings and their difference (continued)

			Schedule 13D	Panel A: F.	iling date – All ev	ents Schedule 13G			Difference	
	$t = Filing \ date$	(t)	(2)	(3)	(v)	(2)	(3)	(t)	(2)	(3)
CAAR	(-45, 0)	0.123***	0.122***	0.122***	-0.002	-0.003	-0.002	0.125***	0.125***	0.124***
	Pos: Neg	392:206	387:209	387:209	299:299	298:300	298:300			
	(0, +45)	$0.046^{**}$	0.045***	$0.043^{***}$	-0.019*	-0.014	-0.014	$0.064^{***}$	$0.059^{***}$	0.057***
	Pos: Neg	338:250	332:256	330:258	290:308	286:310	283:315			
	(-45, +45)	0.161***	0.159***	0.157***	-0.022	-0.018	-0.017	0.182***	0.177***	0.175***
	Pos: Neg	397:201	395:203	393:2005	272:326	283:315	279:319			
SCAAR	(-45, 0)	0.364***	$0.374^{***}$	0.381***	-0.009	-0.017	-0.015	0.373***	0.391***	0.531***
	(0, +45)	0.120***	0.125***	0.116***	-0.032	-0.021	-0.019	0.152***	$0.146^{***}$	0.135***
	(-45, +45)	0.459***	$0.474^{***}$	0.472***	-0.044	-0.042	-0.038	0.503***	0.516***	0.510***
AVAR	(-45, 0)	3.360***	3.584***	3.654***	1.266***	I.333***	1.365	$2.094^{***}$	2.252***	2.289***
	(0, +45)	1.401***	1.530***	1.571***	1.410***	1.511***	1.540	-0.009	0.019***	0.031***
	(-45, +45)	2.381***	2.558***	2.613***	1.340***	1.424***	1.454***	1.041***	1.134***	1.159***
CAAV	(-45, 0)	0.244***			-0.014*			0.258***		
	(0, +45)	0.073***			0.027*			0.046***		
	(-45, +45)	0.309***			0.013			0.296***		
Number of	obs.	598	598	598	598	598	598			
Notes: The ta using firms the fu sionif	the market model (1), the market model (1), to the same stock exc Il sample. Panel B pr icance at the 10%, 59	SCAARs, AVAI ), the Carhart mot change, three-dig esents the results % and 1% level. r	ζ, and CAAV fo del (2), and the F it SIC and size. I for the 'clean' e espectively.	r Schedule 13D fi FF5 model (3). Th In each case, we α vents excluding th	lings, a control sa e control sample i alculate the numb he events where th	mple, and their s formed by us er of positive a ne ten-day discl	difference, 45 d ng the coarsenee ad negative even osure period is v	ays before and af d exact matching tts. Panel A prese tiolated. , **, ***	ter the filing d restricting con ents the results indicate statist	trol for iical

Table 3Abnormal returns, return volatility and volume of Schedule 13D filings, a control<br/>sample and their difference

				Panel B: F	iling date – 'Clea	in' events				
			Schedule 13D			Schedule 13G			Difference	
	t = Filing date	(1)	(2)	(3)	(v)	(2)	(3)	(t)	(2)	(3)
CAAR	(-45, 0)	0.125***	0.126***	0.127***	-0.005	-0.006	-0.006	0.130***	0.132***	0.133***
	Pos: Neg	299:153	295:157	294:158	219:233	210:242	224:228			
	(0, +45)	$0.036^{***}$	0.099***	$0.034^{***}$	-0.010	-0.008	-0.009	$0.047^{***}$	$0.043^{***}$	$0.043^{***}$
	Pos:Neg	245:207	240:212	240:212	222:230	222:230	216:236			
	(-45, +45)	0.153***	0.153***	0.153***	-0.017	-0.016	-0.016	0.170***	0.169***	0.169***
	Pos: Neg	294:158	295:157	295:157	203:249	215:237	208:242			
SCAAR	(-45, 0)	0.369***	$0.384^{***}$	$0.394^{***}$	-0.012	-0.020	-0.020	0.382***	$0.404^{***}$	$0.414^{***}$
	(0, +45)	0.092***	$0.078^{***}$	0.093 * * *	-0.016	-0.011	-0.008	0.109***	0.108***	0.102***
	(-45, +45)	0.435***	0.456***	$0.460^{***}$	-0.032	-0.035	-0.032	0.467***	0.489***	0.492***
AVAR	(-45, 0)	3.273***	3.507***	3.582***	1.274***	1.333***	1.360***	1.999***	2.174***	2.222***
	(0, +45)	1.071***	1.171***	1.206***	1.351***	1.443***	$1.470^{***}$	$-0.28I^{**}$	-0.272***	$-0.264^{***}$
	(-45, +45)	2.171***	2.339***	2.393***	1.314***	1.389***	1.416***	0.857***	$0.950^{***}$	0.977***
CAAV	(-45, 0)	0.262***			-0.015			0.277***		
	(0, +45)	0.059***			0.030			0.029		
	(-45, +45)	0.312***			0.015			0.297***		
Number of	obs.	452	452	452	452	452	452			
Notes: The t using firms the fu	able presents CAAR, the market model (1 to the same stock exu Ill sample. Panel B pr icance at the 10%. 59	, SCAARs, AV/ ), the Carhart m change, three-di resents the resul % and 1% level.	AR, and CAAV todel (2), and th igit SIC and size ts for the 'clean respectively.	for Schedule 13L te FF5 model (3). ' e. In each case, we t' events excluding	filings, a control The control samp e calculate the nu g the events wher	l sample, and the le is formed by 1 mber of positive e the ten-day dis	ir difference, 45 ( asing the coarsend and negative eve closure period is	lays before and a ed exact matching nts. Panel A presi violated. , **, ***	fter the filing d g restricting cor ents the results * indicate statis	ate ttrol for tical

 Table 3
 Abnormal returns, return volatility and volume of Schedule 13D filings, a control sample and their difference (continued)

			Schedule 13D		2177 21 200 Quart 1	Schedule 13G			Difference	
	$t = Filing \ date$	(l)	(2)	(3)	(1)	(2)	(3)	(t)	(2)	(3)
CAAR	(-45, 0)	0.016	0.016	0.012	0.00	0.009	0.007	0.007	0.009	0.005
	Pos: Neg	452:453	450:455	445:460	455:450	450:455	452:453			
	(0, +45)	0.035***	0.039***	$0.040^{***}$	0.011	0.012	0.014	0.025*	0.027**	$0.026^{**}$
	Pos: Neg	512:393	521:384	525:380	464:441	479:426	476:429			
	(-45, +45)	$0.050^{***}$	0.052***	$0.048^{***}$	0.019	0.021	0.020	0.030	0.031	0.017
	Pos: Neg	492:413	488:419	481:426	457:448	468:437	468:437			
SCAAR	(-45, 0)	0.001	0.004	0.000	0.011	0.012	0.010	-0.010	-0.009	-0.010
	(0, +45)	$0.087^{***}$	0.102***	$0.100^{***}$	0.029	$0.034^{*}$	0.036*	0.058**	$0.068^{**}$	$0.064^{**}$
	(-45, +45)	0.082**	0.099***	0.093 **	0.039	0.045	0.045	0.043	0.054	0.048
AVAR	(-45, 0)	1.899***	2.036***	2.085***	1.325***	1.382***	1.413***	0.575***	$0.654^{**}$	0.672**
	(0, +45)	1.553***	<i>I.666</i> ***	1.693***	1.415***	1.518***	1.557***	0.137	0.148	0.136
	(-45, +45)	I.732***	1.858***	1.896***	1.372***	1.452***	1.487***	$0.360^{**}$	0.406**	$0.409^{**}$
CAAV	(-45, 0)	0.157***			0.010			0.147***		
	(0, +45)	0.072***			0.031**			0.042*		
	(-45, +45)	0.226***			0.040			0.186***		
Number of ot	JS.	905	905	905	905	905	905			
Notes: The tab using th firms to the full	le presents CAAR, e market model (1) the same stock exc sample. Panel B pre	SCAARs, AVA , the Carhart me hange, three-di sents the result	AR and CAAV odel (2), and the git SIC and size is for the 'clean	for Schedule 13G e FF5 model (3). 2. In each case, w * events excludin	filings, a control The control samp e calculate the nu g the events wher	sample, and the ole is formed by mber of positive e the ten-day dis	ir difference, 45 c using the coarsen e and negative eve closure period is	lays before and af ed exact matching mts. Panel A press violated. *, **, **	ter the filing de g restricting con ents the results ** indicate	te trol for

Table 4Abnormal returns, return volatility and volume of Schedule 13G filings, a control<br/>sample and their difference

				Panel B: F	iling date – 'Clea	n' events				
			Schedule 13D			Schedule 13G			Difference	
	t = Filing date	(l)	(2)	(3)	(v)	(2)	(3)	(t)	(2)	(3)
CAAR	(-45, 0)	0.008	0.006	0.004	-0.003	-0.002	-0.001	0.011	0.008	0.005
	Pos: Neg	298:299	291:306	290:307	289:308	287:310	290:307			
	(0, +45)	$0.038^{***}$	$0.04I^{***}$	$0.042^{***}$	0.001	0.005	0.005	$0.037^{**}$	$0.036^{**}$	0.037**
	Pos:Neg	336:261	340:257	348:249	302:295	316:281	310:287			
	(-45, +45)	$0.043^{**}$	$0.044^{**}$	$0.041^{**}$	-0.002	0.003	0.005	0.045	0.040	0.036
	Pos: Neg	327:270	328:269	322:275	291:306	299:298	298:299			
SCAAR	(-45, 0)	0.001	-0.000	-0.002	-0.012	-0.011	-0.010	0.013	0.010	0.008
	(0, +45)	$0.090^{***}$	$0.108^{***}$	0.105***	0.007	0.014	0.012	0.083**	$0.094^{**}$	0.093
	(-45, +45)	$0.084^{**}$	$0.098^{***}$	$0.094^{**}$	-0.004	0.005	0.003	0.087	0.093	0.091
AVAR	(-45, 0)	1.742***	1.832***	1.863***	1.280***	1.334***	1.365***	0.463***	0.499***	0.498***
	(0, +45)	$1.674^{***}$	1.771***	1.802***	1.360***	1.464***	1.506***	0.314	0.306	0.296
	(-45, +45)	1.713***	1.807***	1.838***	1.320***	1.340***	<i>I.436***</i>	0.393**	$0.407^{**}$	0.402**
CAAV	(-45, 0)	0.194***			0.019			0.175***		
	(0, +45)	0.105***			0.042**			0.063**		
	(-45, +45)	$0.294^{***}$			0.060*			0.234***		
Number of obs	·	597	597	597	597	597	597			
Notes: The table using the firms to t the full se statistical	presents CAAR, S market model (1), he same stock excl mple. Panel B pre- sionificance at the	SCAARs, AVA the Carhart mo nange, three-dig sents the results	R and CAAV find the control of the c	or Schedule 13G 2 FF5 model (3). ' 1 In each case, we events excluding crively	filings, a control The control samp e calculate the nu g the events wher	sample, and thei le is formed by u mber of positive e the ten-day dis	r difference, 45 d using the coarsene and negative eve closure period is	lays before and af ed exact matching nts. Panel A prese violated. *, **, **	ter the filing da crestricting con ents the results ** indicate	te Itrol for

 Table 4
 Abnormal returns, return volatility and volume of Schedule 13G filings, a control sample and their difference

Figure 1 presents the CAARs for the full sample of Schedule 13D and Schedule 13G filings calculated using the market model for the (-45, 45) window, where t = 0 is the filing date. Schedule 13D targets show much higher CAARs than those of Schedule 13G targeted firms with most of the difference accounted for in the (-45, 0) window.

Figure 2 presents the daily mean-adjusted abnormal volume for the full sample of Schedule 13D and Schedule 13G filings. The daily mean-adjusted abnormal volume is positive and higher from the beginning of the examined window for Schedule 13D events when compared with Schedule 13G events, reaching its highest level 8 days prior the filing date for both filing types.







We further examine hedge fund interventions comparing them with a control sample of non-targeted firms. We compare both Schedule 13D and Schedule 13G filing day events with a control sample of non-targeted firms based on year, exchange, industry (three-digit SIC) and size (market value). The results are presented in Tables 3 and 4 for 13D and 13G filings, respectively.

Table 3 confirms evidence of robust short-term value creation during the pre-and the post-filing periods for the Schedule 13D events while the control sample of non-targeted firms does not present significant positive abnormal returns results during the same period. Abnormal returns differences between the two samples are highly positive and significant. Also, AVAR and volume for 13D filings are larger compared to the control group.

Table 4 presents the results for the Schedule 13G events and for the control sample of non-targeted firms. We find no evidence of short-term value creation for both samples prior to the filing date and for the control sample after the filing date. AVAR and volume for 13G filings are larger compared to the control group but the differences are significant only in the (-45, 0) window.

Figures 3 and 5 present the CAARs for the full sample of Schedule 13D and Schedule 13G firms, respectively, along with the control sample of non-targeted firms using the market model for the (-45, 45) window, where t = 0 is the filing date. Schedule 13D targets experience positive CAARs while those for non-targeted firms remain close to zero for the whole (-45, 45) window. Schedule 13G filings have positive CAARs starting

five days prior to the filing date which remain higher than those of the control sample in the post-filing period.

Figures 4 and 6 present the daily mean-adjusted abnormal volume for the full sample of Schedule 13D and Schedule 13G firms, respectively, along with a control sample of non-targeted firms. The daily mean-adjusted abnormal volume is positive and higher from the beginning of the examined window for Schedule 13D and Schedule 13G events when compared with the control sample of non-targeted firms until about ten days after the filing date.





- Notes: The y axis measures the CAARs in percent for Schedule 13D and a control sample of non-targeted firms calculated using the market model for the period (-45, 45) where t = 0 is the filing date.
- Figure 4 Daily mean-adjusted abnormal volume surrounding the initial filing full sample (see online version for colours)



Note: The y axis measures the daily mean-adjusted abnormal volume for Schedule 13D and a control sample of non-targeted firms for the period (-45, 45) where t = 0 is the filing date.



Figure 5 CAARs surrounding the initial filing – full sample (see online version for colours)

- Note: The y axis measures the CAARs in percent for Schedule 13G and a control sample of non-targeted firms calculated using the market model for the period (-45, 45) where t = 0 is the filing date.
- Figure 6 Daily mean-adjusted abnormal volume surrounding the initial filing full sample (see online version for colours)



Note: The y axis measures the daily mean-adjusted abnormal volume for Schedule 13G and a control sample of non-targeted firms for the period (-45, 45) where t = 0 is the filing date.

#### 4.2 Main event

Our results, from the previous section, indicate the existence of strong positive abnormal returns, return volatility and volume during the pre-filing period of which abnormal returns are significant only for Schedule 13D events. We surmise these findings are indicative of the presence of insider trading in the market given the nature of the 'activist' 13D intervention. However, there is no evidence of significant differences in abnormal returns or abnormal volatility and volume between 13D and 13G targets in the post filing

period. All the measures are positive and significant in this period albeit significantly lower than the corresponding measures in the pre filing period for 13D targeted firms. These findings are indicative of the presence of asymmetric information in the post filing period, notably information associated with target firm insiders who are likely to have better information than other investors in the market about how the firm will be affected by the fund's involvement. As we find no statistical difference between 13D and 13G targeted firms in this period, we conjecture the information asymmetry reflects market beliefs that 13D or 13G targets are undervalued firms. For example, Brav et al. (2008) note that hedge funds typically target 'value' firms, with low market value relative to book value. Of course, whether such firms are truly undervalued time will tell. Yet prices cannot perfectly reflect the information which is available either because is too costly to access (Grossman and Stiglitz, 1980) or because market participants find it too difficult to interpret the information disclosed in the initial filings.

In answering the question, we posited at the start of the paper, the evidence so far suggests that most gains appear to have been made before disclosure. While abnormal returns continue to be positive on average after disclosure, caution needs to be exercised as the immediate post filing returns may be noisy indicators as to where the stock of the targeted firm is heading over the ensuing days or weeks. We turn next to investigate this issue further. Specifically, we examine whether most of the gains in the targeted stock occur before the date where the filer surpasses the 5% threshold and the obligation to disclose the new holdings with the SEC is triggered, the main event date for our purposes. Before the event date, hedge funds still build their stake and since our dataset does not include 'wolf packs', the information about the new ownership is not available to other market participants. After the *event* date the hedge fund must file a Schedule 13D or 13G within 10 days. We restrict our dataset to events where the *event* date lags at least one day from the filing date meaning that the new information could be publicly available at least one day after the event date.

Table 5 presents the daily abnormal returns around the *event* date starting ten days before and ending ten days after the *event* date for the Schedule 13D and Schedule 13G filings for both the full sample (Panel A) and the 'clean' *event* sample (Panel B). The results show the existence of positive abnormal returns for Schedule 13D filings starting from day-5 for the market model, the Carhart and FF5 models. In contrast, the Schedule 13G filings present positive and statistically significant results about three and seven days after the main event date, namely after the filer made the decision to acquire more than 5% of the stock.

We employ the same methodology as in the previous section to calculate CAARs, SCAARs, AVAR and CAAV using as t = 0 as the *event* date. Table 6 presents the results for the Schedule 13D and Schedule 13G filings using three different windows (-10, 0), (-5, 0) and (-1, 0). The results show that Schedule 13D filings perform better in terms of abnormal returns for all examined windows when compared to Schedule 13G filings made by the same hedge funds prior the *event* date. Moreover, abnormal volume is positive and statistically significant for both filing types with Schedule 13D filings having higher abnormal volume than Schedule 13G filings. Differences between the two groups of targeted firms are positive and statistically significant for every examined measure and window.

(I)         (I) <th>(1)</th> <th></th> <th></th> <th></th>	(1)			
Sivent date $AAR$ % $P-value$ $P-value$ $P-value$ $AAR$ % $P-value$ $AAR$ % $P-value$ <		(2)	(3)	
$0.349$ $0.142$ $0.306$ $0.201$ $0.333$ $0.164$ $-0.172$ $-0.052$ $0.732$ $-0.094$ $0.536$ $-0.110$ $0.480$ $-0.190^*$ $0.115$ $0.526$ $0.074$ $0.686$ $0.084$ $0.644$ $0.014$ $0.115$ $0.526$ $0.074$ $0.686$ $0.034$ $0.644$ $0.014$ $0.115$ $0.526$ $0.074$ $0.686$ $0.034$ $0.644$ $0.014$ $0.700^{****}$ $0.005$ $0.682^{***}$ $0.020$ $0.584^{****}$ $0.003$ $0.061$ $0.770^{****}$ $0.002$ $0.682^{***}$ $0.021$ $0.592^{****}$ $0.016$ $0.336^{****}$ $0.012$ $0.662^{****}$ $0.022$ $0.662^{****}$ $0.022$ $0.662^{****}$ $0.022$ $0.052^{************************************$	R % p-value A	IR % p-value	AAR %	p-value
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	.172 0.394 –0	.177 0.384	-0.158	0.440
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	<i>190</i> * 0.097 –0.	196* 0.091	-0.197*	0.093
0.314 $0.105$ $0.327*$ $0.086$ $0.300$ $0.116$ $0.036$ $0.055$ $0.809$ $0.095$ $0.682$ $0.094$ $0.684$ $-0.040$ $0.700***$ $0.002$ $0.682***$ $0.003$ $0.061$ $-0.040$ $0.700***$ $0.002$ $0.683***$ $0.002$ $0.683***$ $0.003$ $0.016$ $0.706***$ $0.016$ $0.388**$ $0.013$ $0.022$ $0.022$ $0.023$ $0.706***$ $0.016$ $0.388**$ $0.011$ $0.682**$ $0.013$ $-0.018$ $0.706***$ $0.009$ $0.683**$ $0.011$ $0.622**$ $0.012$ $0.018$ $0.706***$ $0.009$ $0.683**$ $0.011$ $0.682**$ $0.018$ $0.023$ $1.324**$ $0.000$ $0.423*$ $0.001$ $0.440*$ $0.012$ $0.105$ $0.386**$ $0.000$ $1.379**$ $0.000$ $1.392**$ $0.012$ $0.012$ $0.325***$ $0.000$ <td< td=""><td>014 0.919 0</td><td>012 0.931</td><td>-0.008</td><td>0.956</td></td<>	014 0.919 0	012 0.931	-0.008	0.956
$0.055$ $0.809$ $0.095$ $0.682$ $0.094$ $0.684$ $-0.040$ $0.700^{***}$ $0.002$ $0.685^{***}$ $0.002$ $0.688^{***}$ $0.003$ $0.061$ $0.674^{***}$ $0.002$ $0.685^{***}$ $0.002$ $0.668^{***}$ $0.003$ $0.061$ $0.674^{***}$ $0.016$ $0.368^{***}$ $0.013$ $0.061$ $0.065$ $0.706^{***}$ $0.016$ $0.368^{***}$ $0.013$ $0.013$ $0.018$ $0.706^{***}$ $0.010$ $0.368^{***}$ $0.011$ $0.682^{***}$ $0.012$ $0.105$ $0.736^{***}$ $0.000$ $0.423^{**}$ $0.011$ $0.682^{***}$ $0.012$ $0.105$ $0.736^{***}$ $0.000$ $0.423^{**}$ $0.011$ $0.682^{***}$ $0.012$ $0.105$ $1.324^{***}$ $0.000$ $1.379^{***}$ $0.000$ $1.392^{****}$ $0.012$ $0.022$ $0.335^{****}$ $0.000$ $1.392^{****}$ $0.000$ $0.504^{****}$ $0.016$ $0.022^{**}$ <	036 0.795 0.	018 0.896	-0.018	0.899
$0.700^{***}$ $0.002$ $0.683^{***}$ $0.003$ $0.661$ $0.674^{**}$ $0.023$ $0.662^{***}$ $0.027$ $0.682^{***}$ $0.020$ $0.095$ $0.574^{***}$ $0.013$ $0.662^{***}$ $0.027$ $0.692^{***}$ $0.013$ $0.095$ $0.386^{***}$ $0.016$ $0.368^{***}$ $0.013$ $0.013$ $-0.018$ $0.386^{***}$ $0.010$ $0.562^{***}$ $0.011$ $0.695$ $-0.018$ $0.706^{****}$ $0.009$ $0.685^{***}$ $0.011$ $0.622$ $0.105$ $0.736^{****}$ $0.001$ $0.423^{**}$ $0.011$ $0.632^{***}$ $0.012$ $0.336^{****}$ $0.000$ $1.379^{****}$ $0.000$ $1.379^{****}$ $0.012$ $0.122$ $0.352^{****}$ $0.000$ $1.379^{****}$ $0.000$ $0.544^{****}$ $0.001$ $0.172$ $0.277^{***}$ $0.000$ $0.524^{***}$ $0.016$ $0.212^{**}$ $0.016$ $0.212^{**}$ $0.277^{***}$ $0.001$ $0.277^{$	.040 0.822 0.	000 0.999	-0.006	0.973
$0.674^{**}$ $0.023$ $0.662^{**}$ $0.027$ $0.692^{**}$ $0.020$ $0.095$ $0.386^{**}$ $0.016$ $0.368^{**}$ $0.020$ $0.398^{**}$ $0.013$ $-0.018$ $0.706^{***}$ $0.009$ $0.685^{**}$ $0.011$ $0.682^{**}$ $0.012$ $0.013$ $0.706^{***}$ $0.009$ $0.685^{***}$ $0.011$ $0.682^{***}$ $0.012$ $0.105$ $0.386^{**}$ $0.070$ $0.423^{**}$ $0.011$ $0.682^{***}$ $0.012$ $0.105$ $0.336^{***}$ $0.000$ $0.423^{**}$ $0.001$ $0.440^{***}$ $0.012$ $0.012$ $1.324^{****}$ $0.000$ $1.379^{****}$ $0.000$ $1.392^{****}$ $0.001$ $0.172$ $0.277^{***}$ $0.000$ $1.379^{****}$ $0.000$ $0.524^{****}$ $0.014$ $0.172$ $0.277^{***}$ $0.000$ $0.524^{****}$ $0.001$ $0.277^{***}$ $0.016$ $0.172$ $0.277^{***}$ $0.001$ $0.227^{***}$ $0.001$ $0.212^{**}$ $0.016$ $0.172$ $0.236^{****}$ $0.001$ </td <td>061 0.651 0.</td> <td>084 0.539</td> <td>0.087</td> <td>0.523</td>	061 0.651 0.	084 0.539	0.087	0.523
$0.386^{**}$ $0.016$ $0.368^{**}$ $0.020$ $0.398^{**}$ $0.013$ $-0.018$ $0.706^{***}$ $0.009$ $0.685^{**}$ $0.011$ $0.682^{**}$ $0.012$ $0.105$ $0.706^{***}$ $0.009$ $0.685^{***}$ $0.011$ $0.682^{**}$ $0.012$ $0.105$ $0.386^{*}$ $0.070$ $0.423^{*}$ $0.011$ $0.682^{***}$ $0.012$ $0.105$ $1.324^{***}$ $0.000$ $1.379^{***}$ $0.000$ $1.392^{****}$ $0.001$ $0.045$ $1.324^{***}$ $0.000$ $1.379^{***}$ $0.000$ $1.392^{****}$ $0.001$ $0.015$ $0.277^{***}$ $0.000$ $1.379^{***}$ $0.000$ $0.257^{***}$ $0.014$ $0.172$ $0.277^{***}$ $0.001$ $0.277^{***}$ $0.014$ $0.064$ $0.164$ $0.236^{****}$ $0.007$ $0.277^{***}$ $0.014$ $0.064$ $0.0212^{**}$ $0.303^{*****}$ $0.007$ $0.277^{***}$ $0.014$ $0.212^{**}$ $0.016$ $0.303^{*****}$ $0.001$ $0.277^{***}$ $0.001$ $0.016$ <	095 0.641 0.	122 0.549	0.101	0.617
$0.706^{***}$ $0.009$ $0.683^{**}$ $0.011$ $0.682^{**}$ $0.012$ $0.105$ $0.386^{*}$ $0.070$ $0.423^{*}$ $0.011$ $0.682^{***}$ $0.012$ $0.105$ $1.324^{***}$ $0.070$ $0.423^{*}$ $0.048$ $0.440^{**}$ $0.040$ $-0.239$ $1.324^{***}$ $0.000$ $1.379^{***}$ $0.000$ $1.379^{***}$ $0.001$ $0.045$ $0.277^{**}$ $0.000$ $0.524^{***}$ $0.001$ $0.172$ $0.172$ $0.277^{**}$ $0.000$ $0.524^{***}$ $0.001$ $0.172$ $0.172$ $0.277^{**}$ $0.001$ $0.524^{***}$ $0.001$ $0.172$ $0.172$ $0.277^{**}$ $0.001$ $0.257^{**}$ $0.001$ $0.172$ $0.064$ $0.236^{***}$ $0.001$ $0.271^{***}$ $0.001$ $0.212^{**}$ $0.016$ $0.303^{****}$ $0.001$ $0.271^{***}$ $0.001$ $0.001$ $0.002$ $0.338^{****}$ $0.004$ $0.002$ $0.46^{*}$ <td>.018 0.921 0</td> <td>009 0.963</td> <td>0.005</td> <td>0.977</td>	.018 0.921 0	009 0.963	0.005	0.977
0.386* $0.070$ $0.423*$ $0.048$ $0.440**$ $0.040$ $-0.239$ $1.324***$ $0.000$ $1.379***$ $0.000$ $1.392***$ $0.000$ $0.045$ $0.525***$ $0.000$ $0.524***$ $0.000$ $0.045$ $0.012$ $0.045$ $0.277**$ $0.000$ $0.524**$ $0.001$ $0.172$ $0.054$ $0.277**$ $0.000$ $0.524**$ $0.001$ $0.172$ $0.277**$ $0.000$ $0.524**$ $0.001$ $0.172$ $0.277**$ $0.001$ $0.277**$ $0.044$ $0.064$ $0.236***$ $0.007$ $0.277**$ $0.014$ $0.064$ $0.303***$ $0.007$ $0.276**$ $0.016$ $0.212*$ $0.378***$ $0.007$ $0.276**$ $0.001$ $-0.002$ $0.378***$ $0.004$ $0.406***$ $0.002$ $0.016$ $0.017$	105 0.621 0.	144 0.498	0.149	0.484
$I.324^{***}$ $0.000$ $I.379^{***}$ $0.000$ $I.392^{***}$ $0.000$ $0.045$ $0.525^{***}$ $0.000$ $0.524^{***}$ $0.000$ $0.001$ $0.172$ $0.277^{***}$ $0.003$ $0.227^{*}$ $0.001$ $0.172$ $0.277^{***}$ $0.033$ $0.227^{*}$ $0.004$ $0.064$ $0.286^{***}$ $0.007$ $0.277^{**}$ $0.014$ $0.064$ $0.237^{***}$ $0.007$ $0.277^{**}$ $0.016$ $0.172$ $0.386^{***}$ $0.007$ $0.277^{**}$ $0.014$ $0.016$ $0.015$ $0.337^{****}$ $0.007$ $0.276^{***}$ $0.016$ $0.017^{*}$ $0.022$ $0.378^{****}$ $0.004$ $0.406^{***}$ $0.002$ $0.010$ $-0.002$	.239 0.364 –0	.198 0.455	-0.168	0.527
0.525** $0.000$ $0.524**$ $0.001$ $0.172$ $0.277**$ $0.033$ $0.227*$ $0.001$ $0.172$ $0.277**$ $0.033$ $0.227*$ $0.078$ $0.257**$ $0.044$ $0.064$ $0.286***$ $0.007$ $0.272**$ $0.014$ $0.267**$ $0.016$ $0.212*$ $0.386***$ $0.007$ $0.272**$ $0.014$ $0.271***$ $0.016$ $0.172$ $0.303****$ $0.007$ $0.276**$ $0.014$ $0.274**$ $0.016$ $0.175$ $0.378***$ $0.004$ $0.406***$ $0.002$ $0.415***$ $0.001$ $-0.002$	045 0.861 0.	075 0.771	0.080	0.754
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	0.222 0.222 0.	172 0.230	0.149	0.297
0.286*** 0.007 0.272** 0.014 0.271*** 0.016 0.212* 0.303*** 0.007 0.276** 0.018 0.254** 0.037 0.175 0.378*** 0.004 0.406*** 0.002 0.415*** 0.001 -0.002	064 0.634 0.	089 0.509	0.056	0.675
0.303*** 0.007 $0.276**$ 0.018 $0.254**$ 0.037 0.175 0.378*** 0.004 $0.406***$ 0.002 $0.415***$ 0.001 $-0.002$	0.085 0.085	178 0.153	0.188	0.129
0.378*** 0.004 $0.406***$ 0.002 $0.415***$ 0.001 $-0.002$	175 0.130 0.	155 0.182	$0.194^{*}$	0.094
	.002 0.987 0.	028 0.811	0.024	0.836
160.0- / 06.0 061.0 007.0 021.0 0/0.0 v0.0 v0.0	.091 0.448 –0	.097 0.415	-0.077	0.524
0.447*** 0.000 0.409*** 0.001 0.408*** 0.001 0.290**	90** 0.043 0.	265* 0.059	0.261*	0.066
0.384*** 0.005 0.348** 0.014 0.331** 0.018 0.270**	70** 0.015 0.3.	29*** 0.003	$0.294^{***}$	0.008
0.396** 0.019 0.431** 0.013 0.433** 0.010 0.173	0.102 0.102 0.	147 0.154	0.114	0.265
0.244* 0.088 0.154 0.281 0.142 0.326 0.193*	93* 0.068 0.	179 0.101	0.202	0.065

Table 5Daily abnormal returns around the event date

E. Karpouzis et al.

298

			Schedu	le I3D					Schedule	13G		
	(l)		(2)	-	(3,		$\mathcal{U}$	(	(2)		(3)	
t = Event date AAI	R % F	o-value	AAR %	p-value	AAR %	p-value	AAR %	p-value	AAR %	p-value	AAR %	p-value
-10 0.5	02	0.100	0.466	0.129	0.510*	0.097	-0.224	0.432	-0.234	0.413	-0.214	0.454
-0-	024	0.882	-0.078	0.639	-0.107	0.527	-0.236	0.113	-0.225	0.134	-0.218	0.151
-8 0.0	65	0.774	0.015	0.947	0.011	0.959	0.030	0.861	0.018	0.918	0.011	0.945
-7 0.3	44	0.114	0.351	0.102	0.315	0.143	0.128	0.464	0.123	0.476	0.088	0.614
-0.0	043	0.867	-0.010	0.969	-0.009	0.973	-0.231	0.136	-0.214	0.182	-0.222	0.160
-5 0.77	0***	0.005	0.777***	0.005	$0.740^{***}$	0.007	-0.021	0.897	0.018	0.914	0.025	0.881
-4 0.77	**1	0.044	0.797**	0.040	$0.817^{**}$	0.035	-0.224	0.221	-0.186	0.305	-0.225	0.217
-3 0.3	*00	0.061	0.279*	0.079	0.300*	0.061	0.034	0.875	0.087	0.686	0.065	0.762
-2 0.65	**[]	0.018	0.638**	0.020	$0.620^{**}$	0.025	0.297	0.317	0.302	0.308	0.315	0.288
-1 0.4.	24*	0.086	0.473*	0.058	0.503**	0.044	-0.583 ***	0.000	-0.565 ***	0.000	$-0.510^{***}$	0.001
0 1.15	8***	0.003	1.239***	0.002	1.247***	0.002	0.347	0.310	0.396	0.243	0.400	0.237
1 0.47	7***	0.006	0.492***	0.005	0.459**	0.010	0.142	0.383	0.192	0.245	0.163	0.321
2 0.34.	S***	0.009	0.269 **	0.040	0.289**	0.027	0.032	0.853	0.083	0.622	0.049	0.771
3 0.38.	3***	0.002	0.391***	0.003	$0.394^{***}$	0.003	0.146	0.335	0.129	0.395	0.138	0.360
4 0.1	92	0.100	0.191	0.120	0.148	0.260	0.205	0.117	0.179	0.170	0.200	0.121
5 0.1	70	0.196	0.211	0.103	0.231*	0.074	0.022	0.854	0.018	0.875	0.026	0.820
6 0.2	093	0.103	0.211	0.195	0.175	0.291	-0.020	0.868	-0.004	0.971	0.000	0.995
7 0.31	**	0.024	0.325**	0.018	0.325**	0.019	0.343**	0.012	0.315**	0.020	0.277**	0.039
8 0.3	02*	0.066	0.269	0.109	0.271	0.106	0.121	0.323	0.181	0.134	0.176	0.145
9 0.2	:55	0.117	0.249	0.126	0.276*	0.086	0.262**	0.048	0.222*	0.080	0.179	0.155
10 0.1	90	0.235	0.085	0.590	0.076	0.633	0.243**	0.047	0.293**	0.018	0.335***	0.007
Notes: The table presen model (1), the C the events where	tts averag arhart m the ten-	ge abnorma Iodel (2), ar dav disclos	al stock return nd the FF5 mc sure period is	s (AAR) ten del (3). Panε violated. *. *	days before an al A presents th **. *** indicate	d after the ev the results for the statistical si	/ent date for Sch the full sample.	hedule 13D and Panel B prese e 10%. 5% and	d Schedule 13G nts results for th 11% level. resp	f filings using ne 'clean' ev ectivelv.	g the market ents excluding	

Table 6Abnormal returns, return volatility and volume of Schedule 13D filings, Schedule<br/>13G filings and their difference

			Schedule 13D			Schedule 13G			Difference	
	t=Filing Date	(t)	(2)	(3)	(t)	(2)	(3)	(t)	(2)	(3)
CAAR	(-10, 0)	$0.049^{***}$	0.050***	$0.050^{***}$	-0.007	-0.005	-0.005	0.056***	$0.054^{***}$	0.055***
	Pos.:Neg.	254:226	260:220	261:219	285:379	297:367	303:361			
	(-5, 0)	$0.04I^{***}$	$0.042^{***}$	0.042***	-0.002	0.001	0.001	0.042***	0.042***	0.041***
	Pos.: Neg.	260:220	266:214	274:206	307:357	316:348	318:350			
	(-1, 0)	0.016***	0.017***	$0.018^{***}$	-0.002	-0.002	-0.001	$0.018^{***}$	$0.019^{***}$	0.019***
	Pos.: Neg.	255:225	257:223	272:208	319:345	316:348	319:345			
SCAAR	(-10, 0)	0.138***	0.145***	$0.147^{***}$	-0.027*	-0.023	-0.024	0.165***	0.169***	0.176***
	(-5, 0)	0.115***	0.123***	0.124***	-0.008	-0.004	-0.003	0.123***	0.127***	0.127***
	(-1, 0)	$0.043^{***}$	$0.047^{***}$	$0.048^{***}$	-0.002	-0.001	0.001	0.045***	$0.048^{***}$	0.047***
AVAR	(-10, 0)	5.771***	6.242***	6.360***	2.731***	2.891***	2.939***	3.041***	3.351***	3.421***
	(-5, 0)	6.809***	7.407***	7.545***	3.192***	3.376***	3.438***	3.617***	4.031***	4.107***
	(-1, 0)	8.027***	8.670***	8.812***	4.471***	4.772***	4.873***	3.556*	3.898*	3.939*
CAAV	(-10, 0)	0.152***			0.097***			0.055***		
	(-5, 0)	0.099***			0.073***			0.026*		
	(-1, 0)	$0.048^{***}$			$0.040^{***}$			0.008		
Number o	f obs.	480	480	480	664	664	664			
Notes: The usin	table presents CAAF g the market model (	s) SCAARs, AV	AR, and CAAV f del (2), and the F	or Schedule 13D, S F5 model (3). In ea	schedule 13G filing ach case, we calcul	s, and their diffe ate the number of	erence, (-10, -5, - of positive and ne	-1) days before gative events. P	the event date anel A preser	ts

Table 6Abnormal returns, return volatility and volume of Schedule 13D filings, Schedule<br/>13G filings and their difference (continued)

				, T	<sup>9</sup> anel A: Event	date – all eve	nts					
			Schedule	e 13D					Schedu	le 13G		
	'n	-	(2)		(3)		3	6	3	(2	(3,	
t = Event date	AAR %	p-value	AAR %	p-value	AAR %	p-value	AAR %	p-value	AAR %	p-value	AAR %	p-value
-10	0.138	0.484	0.097	0.627	0.181	0.470	0.195	0.090	0.164	0.167	0.092	0.466
6	-0.045	0.777	-0.053	0.742	-0.014	0.938	-0.093	0.476	-0.057	0.652	-0.002	0.993
-8	0.134	0.486	0.068	0.729	-0.018	0.942	0.054	0.721	0.068	0.651	0.124	0.500
L	0.354*	0.081	0.374*	0.061	0.364	0.102	0.092	0.495	0.067	0.615	-0.005	0.977
9-	0.109	0.644	0.175	0.470	0.066	0.813	-0.171	0.139	-0.137	0.244	-0.186	0.155
<b>S</b> <sup></sup>	0.765***	0.001	$0.747^{**}$	0.002	$0.848^{***}$	0.004	0.209	0.161	0.214	0.135	0.127	0.364
4	0.625**	0.046	0.622*	0.051	$0.794^{*}$	0.054	0.211	0.249	0.186	0.295	0.246	0.255
- S	$0.387^{**}$	0.023	0.352**	0.038	0.278*	0.099	0.077	0.640	0.029	0.861	0.014	0.938
-2	0.759***	0.008	0.746***	0.008	0.712**	0.013	0.042	0.757	0.082	0.552	0.063	0.708
-1	0.436*	0.054	$0.484^{**}$	0.035	0.600*	0.025	-0.177	0.215	-0.168	0.241	-0.224	0.176
0	1.131***	0.001	1.184***	0.000	$I.009^{***}$	0.007	0.177	0.216	0.200	0.171	0.226	0.150
1	0.421***	0.001	0.450***	0.000	$0.426^{***}$	0.004	0.017	0.884	0.017	0.889	-0.127	0.349
2	0.278**	0.033	0.211	0.100	0.277**	0.041	0.055	0.653	-0.022	0.865	0.056	0.690
3	0.287***	0.009	$0.334^{***}$	0.004	0.453***	0.002	-0.035	0.772	0.046	0.705	0.074	0.561
4	0.277**	0.019	0.277	0.025	0.145	0.296	-0.012	0.922	-0.018	0.883	-0.124	0.361
5	0.399***	0.004	0.448***	0.001	0.261**	0.049	0.016	0.885	0.066	0.567	-0.041	0.763
6	0.214	0.138	0.195	0.191	0.211	0.234	-0.076	0.525	-0.032	0.786	-0.086	0.558
7	$0.504^{***}$	0.000	0.463***	0.000	0.333**	0.026	0.265*	0.050	0.280*	0.039	$0.264^{*}$	0.096
8	0.379***	0.009	0.357**	0.018	0.280	0.115	-0.006	0.960	-0.067	0.585	-0.022	0.880
6	0.419**	0.010	0.429**	0.017	0.311*	0.068	0.059	0.658	0.102	0.475	0.151	0.389
10	0.291*	0.054	0.198	0.201	0.140	0.425	-0.118	0.430	-0.229	0.136	-0.080	0.575
Notes: The table p (1), the Car exchange, t ten-day dise	resents averag hart model (2) hree-digit SIC closure period	ge abnormal s ), and the FF: ) and size. Pa   is violated. *	tock returns (Az 5 model (3). The nel A presents tl *, **, *** indical	AR) 10 days 1 c control sam he results for te statistical s	before and after ole is formed by the full sample	the event dat y using the cc . Panel B pre-	e for Schedul arsened exac sents results f and 1% level,	e 13D and a c t matching res or the 'clean' respectively.	control sampl stricting cont events exclu	le using the m rol firms to th ding the even	larket model le same stock ts where the	

**Table 7**Daily abnormal returns around the event date

			Schedul	e 13D	I anel D. LVen	anne - crear	evenus		Schedu	le 13G		
	Ú	(	(2)		(3)		(I)		0	(	(3)	
t = Event date	AAR %	p-value	AAR %	p-value	AAR %	p-value	AAR %	p-value	AAR %	p-value	AAR %	p-value
-10	0.209	0.394	0.166	0.506	0.181	0.470	0.087	0.483	0.084	0.509	0.093	0.466
6	0.000	0.998	-0.022	0.899	-0.014	0.938	-0.005	0.974	0.017	0.900	-0.001	0.993
8-	0.055	0.819	-0.015	0.950	-0.018	0.942	0.152	0.402	0.136	0.454	0.124	0.500
L	0.358	0.110	$0.384^{*}$	0.084	0.364	0.102	0.019	0.901	-0.006	0.968	-0.005	0.977
9-	0.022	0.936	0.076	0.784	0.066	0.813	-0.238 **	0.056	-0.184	0.149	-0.186	0.155
-5	0.852***	0.003	0.885***	0.003	$0.848^{***}$	0.004	0.118	0.422	0.147	0.300	0.128	0.364
4	0.747*	0.065	0.778*	0.059	$0.794^{*}$	0.054	0.226	0.311	0.198	0.359	0.247	0.255
$\tilde{c}^{-}$	0.291*	0.080	0.250	0.138	0.278*	0.099	0.129	0.504	0.054	0.780	0.015	0.938
-2	0.732**	0.010	0.749***	0.008	0.712**	0.013	-0.054	0.739	0.021	0.898	0.064	0.708
-1	0.477*	0.070	0.551**	0.040	$0.600^{**}$	0.025	-0.297*	0.069	-0.274*	0.094	-0.224	0.176
0	0.919**	0.015	1.011***	0.007	$1.009^{***}$	0.007	0.144	0.332	0.185	0.225	0.226	0.150
1	0.370**	0.010	$0.408^{***}$	0.005	0.426***	0.004	-0.114	0.365	-0.106	0.412	-0.127	0.349
2	$0.354^{***}$	0.009	0.267**	0.048	0.277**	0.041	0.176	0.208	0.085	0.552	0.057	0.690
3	0.395***	0.002	0.462***	0.001	0.453***	0.002	-0.018	0.891	0.064	0.621	0.074	0.561
4	0.176	0.153	0.177	0.171	0.145	0.296	-0.001	0.992	-0.052	0.703	-0.124	0.361
5	0.166	0.218	0.232*	0.080	0.261**	0.049	-0.083	0.537	-0.016	0.904	-0.041	0.763
9	0.237	0.161	0.224	0.192	0.211	0.234	-0.137	0.315	-0.084	0.538	-0.085	0.558
7	0.355**	0.016	0.342**	0.020	$0.333^{***}$	0.026	0.163	0.299	0.225	0.158	0.265*	0.096
8	0.296*	0.087	0.280	0.117	0.280	0.115	0.057	0.678	0.010	0.941	-0.022	0.880
6	0.291*	0.089	0.283	0.104	$0.311^{*}$	0.068	0.105	0.484	0.127	0.433	0.152	0.389
10	0.239	0.154	0.141	0.412	0.140	0.425	-0.008	0.949	-0.100	0.473	-0.079	0.575
Notes: The table I (1), the Ca exchange, ten-day dis	resents avera rhart model ( three-digit SI	tige abnormal 2), and the FF C and size. Pa d is violated	stock returns ( <sup>35</sup> model (3). T anel A presents * ** *** indi	AAR) 10 day The control se s the results f	/s before and af umple is formed for the full sam	fter the event I by using the ple. Panel B p	date for Sched coarsened exi presents results % and 1% leve	ule 13D and a tot matching 1 for the 'clear d'respectivel	a control sam estricting con n' events excl	ple using the 1 ntrol firms to t uding the eve	market model he same stock nts where the	
icin-day uri	sciosure perio	d is viulated.	, , , , , пип	Calc statistic	al significance	dl UIC 1070, J	20 GUU 1 20 10 VG	il, Fespecuver				

 Table 7
 Daily abnormal returns around the event date (continued)

			Schedule 13D	Panel A: Fili	ng date – all eveni	s Schedule 13G			Difference	
-	t = Filing date	(t)	(2)	(3)	(t)	(2)	(3)	<i>(t)</i>	(2)	(3)
CAAR	(-10, 0)	$0.048^{***}$	$0.048^{***}$	$0.049^{***}$	0.006	0.007	0.007	0.042***	0.042***	0.043***
	Pos.:Neg.	321:274	340:255	332:263	231:304	293:302	303:292			
	(-5, 0)	$0.04I^{***}$	$0.04I^{***}$	$0.042^{***}$	0.005	0.005	0.006	0.036***	$0.036^{***}$	0.036***
	Pos.: Neg.	319:279	327:268	341:254	288:307	306:289	302:293			
	(-1, 0)	$0.016^{***}$	0.017***	$0.017^{***}$	0.000	0.000	0.001	$0.016^{***}$	$0.016^{***}$	$0.016^{***}$
	Pos.: Neg.	316:279	329:266	335:260	280:315	294:301	295:300			
SCAAR	(-10, 0)	0.142***	0.146***	0.150***	0.011	0.011	0.012	0.132***	0.136***	0.138***
	(-5, 0)	0.123***	0.127***	0.131***	0.007	0.006	0.008	0.116***	0.120***	0.123***
	(-1, 0)	0.052***	$0.054^{***}$	$0.056^{***}$	-0.004	-0.004	-0.003	0.056***	0.058***	0.059***
AVAR	(-10, 0)	5.812***	6.201***	6.274***	<i>I.483***</i>	1.532***	1.574***	4.329***	4.670***	4.700***
	(-5, 0)	$7.670^{***}$	8.199***	8.276***	$I.689^{***}$	1.737***	1.775***	5.981***	6.462***	6.501***
	(-1, 0)	$10.810^{***}$	11.357***	11.370***	1.691***	1.755***	1.794***	9.118***	9.601***	9.576***
CAAV	(-10, 0)	0.139**			0.001			0.139***		
	(-5, 0)	0.092***			0.002			$0.090^{***}$		
	(-1, 0)	$0.047^{***}$			0.002			$0.046^{***}$		
Number of	obs.	595	595	595	595	595	595			
Notes: The t date - contr result statist	able presents CAAR using the market mo ol firms to the same is for the full sample rical significance at	ts, SCAARs, AV/ odel (1), the Carhan stock exchange, th 2. Panel B presents the 10%. 5% and	AR, and CAAV fr rt model (2), and hree-digit SIC and i results for the 'c	or Schedule 13D fili the FF5 model (3). ' d size. In each case, lean' events exclud ivelv.	ngs, a control sam The control sampl , we calculate the ing the events whe	ple, and their di e is formed by u number of positi rre the ten-day d	fference, (-10, -5 sing the coarsene ve and negative e isclosure period i	(, -1) days befor d exact matching vents. Panel A p s violated. *, **,	e the event g restricting presents the , *** indicate	

Table 8Abnormal returns, return volatility and volume of Schedule 13D filings, a control<br/>sample and their difference

				Panel B: Filing	g date – 'clean' ev	ents				
			Schedule 13D			Schedule 13G			Difference	
-	t=Filing Date	(t)	(2)	(3)	(t)	(2)	(3)	(t)	(2)	(3)
CAAR	(-10, 0)	$0.047^{***}$	$0.048^{***}$	$0.048^{***}$	0.003	0.004	0.005	$0.044^{***}$	$0.044^{***}$	$0.043^{***}$
	Pos.:Neg.	239:212	259:192	253:198	214:237	219:332	229:335			
	(-5, 0)	$0.040^{***}$	0.042***	$0.042^{***}$	0.003	0.003	0.005	$0.038^{***}$	0.039***	0.037***
	Pos.: Neg.	241:210	250:201	256:195	213:238	230:221	230:221			
	(-1, 0)	$0.014^{***}$	0.016***	$0.016^{***}$	-0.002	-0.001	0.000	0.016***	0.017***	$0.016^{***}$
	Pos.: Neg.	238:212	249:202	254:197	199:252	218:233	221:230			
SCAAR	(-10, 0)	0.133***	0.142***	$0.144^{***}$	0.004	0.006	0.008	0.129***	0.136***	0.136***
	(-5, 0)	0.115***	$0.124^{***}$	0.126***	0.003	0.004	0.006	0.112***	0.119***	0.120***
	(-1, 0)	$0.040^{***}$	$0.044^{***}$	0.045***	-0.008	-0.007	-0.006	$0.048^{***}$	$0.05I^{***}$	0.051***
AVAR	(-10, 0)	$5.60I^{***}$	6.050***	6.135***	1.522***	1.561***	1.600***	4.079***	4.489***	4.535***
	(-5, 0)	6.925***	7.524***	7.623***	1.789***	<i>I.836</i> ***	1.876***	5.137***	5.689***	5.747***
	(-1, 0)	7.875***	8.494***	8.531***	1.749***	<i>I.807</i> ***	1.853***	6.126***	6.687***	6.678***
CAAV	(-10, 0)	0.153***			-0.001			0.155***		
	(-5, 0)	$0.100^{***}$			0.000			$0.100^{***}$		
	(-1, 0)	$0.048^{***}$			0.001			$0.047^{***}$		
Number of	obs.	451	451	451	451	451	451			
Notes: The t date 1 contr result statisi	able presents CAAR using the market mo ol firms to the same is for the full sample	ts, SCAARs, AV / odel (1), the Carhan stock exchange, ti 2. Panel B presents the 10%. 5% and	AR, and CAAV fr rt model (2), and hree-digit SIC an 5 results for the 'c	or Schedule 13D fil the FF5 model (3). d size. In each case !ean' events exclud ivelv.	ings, a control san The control sampl , we calculate the 1 ling the events who	pple, and their di e is formed by u number of positi are the ten-day d	ifference, (-10, -5 ising the coarsene ve and negative e lisclosure period ii	5, -1) days befor d exact matchin vents. Panel A F s violated. *, **	e the event g restricting presents the , *** indicate	

Table 8Abnormal returns, return volatility and volume of Schedule 13D filings, a control<br/>sample and their difference (continued)

Table 7 reports daily abnormal returns around the *event* (t = 0) date starting ten days before and ending ten days after the *event* date for the Schedule 13D and a control sample of non-targeted firms for both the full sample (Panel A) and the 'clean' *events* sample (Panel B). Similar to Table 5, the results in Table 7 show the existence of positive abnormal returns for Schedule 13D filings starting from day-7 for the market model and the Carhart model while the control group of firms does not present significant abnormal returns.

Table 8 provides a comparison of cumulative results between the Schedule 13D filings and the control group of non-targeted firms. CAARs, SCAARs and CAAVs for the control sample are close to zero and statistically insignificant. Notably, the differences between the Schedule 13D filings and the control group of firms are highly positive and statistically significant confirming short-term value creation prior to the event date.

We then compare the Schedule 13G filings with a control group of non-targeted firms. The daily abnormal returns are presented in Table 9 where the results are mixed for the full sample and the 'clean' sample of Schedule 13G targeted firms with some evidence of positive abnormal returns close to the filing date mainly for 13G firms.

In Table 10, we provide a comparison between the Schedule 13G filings and the control group of non-targeted firms cumulative results. We find that Schedule 13G filings and target firms have slightly negative CAARs and SCAARs which are statistically significant in the (-10, 0) window. AVAR and volume are significantly greater for 13G firms.

A question that arises is whether the abnormal returns and high level of turnover that we observe prior to the main (trigger) event date are indeed created by hedge fund activists' exploiting private information rather than recognising these firms are already special and hence attract considerable investor interest.<sup>12</sup> To address this question, we follow Wong (2020) by asking the same question, namely, whether 'many investors would independently and spontaneously decide to accumulate shares in the target firms on the same day.' To do so we calculate the total trading volume for the 60-day window around the trigger event date for both the 640 Schedule 13D and 1,011 Schedule 13G filings. The total turnover is calculated for each campaign day and is presented as a percentage of normal trading volume as follows:

$$Turnover = \frac{Vol_{i,t}}{Avg\left(Vol_{i,t-120}...Vol_{i,t-60}\right)}$$
(8)

where the normal trading volume is calculated using the average trading volume from day-120 to day-60 prior the event date. If the ratio equals one, we conclude there is no abnormal turnover on that date. As shown in Figures 7 and 8, the turnover on the event date is about 673% and 525% for Schedule 13D and Schedule 13G filings, respectively. As Wong (2020) notes "since the trigger date is not publicly observable until the 13D [or 13G] is filed, the high level of turnover cannot be a public reaction to the activist's campaign". Note that the endogeneity concern in Wong's (2020) study is that 'pack members are merely joining campaigns that are more likely to be successful, rather than causing such success'. Our dataset does not contain 'wolf pack' Schedule 13D filings albeit we cannot exclude the possibility that wolf packs may be formed informally.

						)		
	(3)		(1)		(2)		(3)	
1AR % p-value	AAR %	p-value	AAR %	p-value	AAR %	p-value	AAR %	p-value
-0.029 0.897	-0.012	0.958	-0.183	0.139	-0.153	0.211	-0.163	0.181
-0.189 0.142	-0.211	0.104	-0.189	0.108	-0.118	0.314	-0.132	0.253
0.748 0.748	0.026	0.861	-0.403***	0.000	-0.415***	0.000	-0.435***	0.000
0.041 0.792	-0.012	0.940	-0.022	0.858	-0.040	0.750	-0.036	0.776
0.010 0.959	-0.032	0.870	-0.045	0.716	-0.052	0.675	-0.067	0.596
.060 0.684	0.072	0.623	-0.211*	0.079	-0.297	0.017	-0.285 **	0.025
084 0.702	0.088	0.687	0.096	0.419	0.091	0.440	0.066	0.576
0.847 0.847	0.026	0.898	0.039	0.784	0.016	0.911	0.018	0.895
21 0.604	0.127	0.586	0.001	0.993	0.064	0.565	0.040	0.716
05 0.486	-0.170	0.565	-0.051	0.650	-0.021	0.851	-0.024	0.835
16 0.955	0.020	0.940	-0.154	0.169	-0.163	0.140	-0.157	0.159
)** 0.038	0.273*	0.065	0.000	0.998	-0.042	0.741	-0.063	0.635
96 0.518	0.085	0.567	-0.157	0.193	-0.069	0.576	-0.099	0.430
8 0.171	0.189	0.168	-0.133	0.266	-0.141	0.240	-0.203*	0.094
0.381	0.119	0.308	-0.168	0.151	-0.169	0.148	-0.117	0.325
97 0.451	0.067	0.605	0.167	0.213	0.148	0.274	0.110	0.411
29 0.325	-0.105	0.431	0.004	0.980	0.048	0.752	0.074	0.621
57* 0.081	0.266*	0.085	0.462	0.104	0.493*	0.084	0.514*	0.071
5*** 0.003	0.319***	0.007	0.074	0.501	0.098	0.373	0.059	0.593
17 0.299	0.093	0.412	-0.075	0.496	-0.080	0.462	-0.060	0.587
56** 0.025	0.284	0.013	0.069	0.519	0.039	0.727	0.028	0.803

E. Karpouzis et al.

306

					Panel B: Eveni	t date – 'clea	n' events					
			Scheduli	e 13D					Schedule	13G		
	(v)		(2)		(3)		Ú)		(2)		(3)	
t = Event date	AAR %	p-value	AAR %	p-value	AAR %	p-value	AAR %	p-value	AAR %	p-value	AAR %	p-value
-10	-0.111	0.725	-0.087	0.784	-0.069	0.830	-0.016	0.911	-0.012	0.934	0.002	0.988
6-	-0.242	0.144	-0.244	0.145	-0.257	0.127	-0.196	0.143	-0.118	0.369	-0.124	0.342
-8	0.037	0.844	0.030	0.877	0.012	0.947	$-0.416^{***}$	0.001	$-0.428^{***}$	0.001	-0.441 ***	0.001
L	0.126	0.514	0.170	0.379	0.115	0.552	-0.102	0.486	-0.115	0.435	-0.124	0.410
9–	-0.226	0.175	-0.188	0.285	-0.198	0.253	0.059	0.713	0.052	0.744	0.049	0.760
-5	-0.039	0.827	-0.019	0.917	-0.005	0.979	-0.281*	0.063	-0.360 **	0.021	$-0.319^{**}$	0.047
4	-0.314	0.103	-0.283	0.141	-0.296	0.123	0.084	0.535	0.085	0.527	0.040	0.769
-3	0.099	0.676	0.135	0.570	0.097	0.683	0.136	0.478	0.116	0.534	0.111	0.555
-2	0.384	0.241	0.379	0.249	0.395	0.231	0.131	0.309	0.161	0.232	0.150	0.273
-1	-0.598***	0.000	$-0.607^{***}$	0.000	$-0.551^{***}$	0.001	-0.135	0.332	-0.120	0.392	-0.097	0.496
0	0.350	0.327	0.402	0.259	0.400	0.263	-0.186	0.150	-0.224*	0.080	-0.221*	0.088
1	0.156	0.380	0.236	0.207	0.193	0.297	0.080	0.585	0.020	0.897	-0.013	0.935
2	0.018	0.926	0.106	0.572	0.086	0.649	-0.189	0.206	-0.104	0.496	-0.15	0.326
3	0.089	0.588	0.083	0.618	0.085	0.609	-0.130	0.359	-0.132	0.348	-0.193	0.175
4	0.156	0.222	0.155	0.226	0.171	0.181	-0.032	0.828	-0.005	0.972	0.06	0.678
5	0.012	0.927	0.062	0.616	0.033	0.788	-0.057	0.669	-0.079	0.559	-0.112	0.405
9	-0.063	0.618	-0.045	0.725	-0.027	0.840	-0.177	0.366	-0.126	0.520	-0.112	0.569
7	0.335**	0.018	$0.292^{**}$	0.038	0.240*	0.087	0.239	0.150	0.303	$0.068^{*}$	$0.312^{*}$	0.060
8	0.130	0.318	0.190	0.146	0.178	0.177	-0.001	0.996	0.019	0.878	0.014	0.905
9	0.233	0.108	0.158	0.259	0.116	0.410	-0.085	0.521	-0.089	0.487	-0.091	0.487
10	0.273**	0.029	0.314**	0.014	0.362***	0.005	-0.002	0.985	0.012	0.925	-0.008	0.953
Notes: The table (1), the C <sup>i</sup> events wh	presents avera urhart model ( are the ten-da	ige abnormal 2), and the FF y disclosure p	stock returns ( <sup>7</sup> 5 model (3). F period is violate	AAR) ten da anel A presei ed. *, **, ***	ys before and a nts the results f indicate statist	fter the even for the full sa tical significa	t date for Sche mple. Panel B unce at the 10%	dule 13G and presents resu 6, 5% and 1%	a control samp lts for the 'clea	ole using the m' events, ex vely.	market model celuding the	

Table 9
 Daily abnormal returns around the event date (continued)

			Schedule 13G			Control sample			Difference	
	t = Filing date	(i)	(2)	(3)	(t)	(2)	(3)	(1)	(2)	(3)
CAAR	(-10, 0)	-0.002	-0.000	-0.001	$-0.011^{***}$	-0.011 **	$-0.012^{***}$	0.009	0.011	0.011
	Pos: Neg	398:484	411:481	413:479	394:498	394:498	399:493			
	(-5, 0)	-0.000	0.001	0.002	-0.003	-0.003	-0.003	0.003	0.004	-0.001
	Pos: Neg	425:467	412:480	415:477	404:488	410:482	410:482			
	(-1, 0)	-0.002	-0.002	-0.002	-0.002	-0.002	-0.002	-0.001	-0.000	-0.000
	Pos: Neg	410:482	430:452	437:445	400:492	398:494	407:485			
SCAAR	(-10, 0)	-0.021	-0.016	-0.016	$-0.024^{***}$	$-0.025^{***}$	-0.026	0.003	0.009	0.010
	(-5, 0)	-0.008	-0.004	-0.002	-0.009	-0.00-	-0.009	-0.002	0.005	0.007
	(-1, 0)	-0.004	-0.002	-0.001	-0.006	-0.005	-0.005	0.002	0.004	0.004
AVAR	(-10, 0)	3.428***	3.767***	3.883***	1.298***	1.353***	1.384***	2.130***	2.414**	2.499**
	(-5, 0)	4.507***	5.023***	5.203***	1.373***	1.442***	$I.474^{***}$	3.134**	3.581*	3.729*
	(-1, 0)	8.470*	9.823*	10.292***	$I.340^{***}$	1.409***	1.441***	7.131	8.414	8.851
CAAV	(-10, 0)	0.082***			0.007			0.075***	0.075***	
	(-5, 0)	$0.064^{***}$			0.006			0.058***	0.058***	
	(-1, 0)	$0.034^{***}$			0.004*			0.030***	$0.030^{***}$	
Number o	f obs.	892	892	892	892	892	892			
Votes: The the 1	table presents CA/ market model (1), t	ARs, SCAARs, A he Carhart mode	VVAR, and CAAV 1 (2), and the FF5	/ for Schedule 13G model (3). In each	i, a control sample case, we calculat	e, and their differ e the number of	rence, $(-10, -5, -$ positive and nega	<ol> <li>days before th tive events. Pane</li> </ol>	ie event date us el A presents th	e no.

Table 10Abnormal returns, return volatility and volume of Schedule 13G filings, a control<br/>sample and their difference

			Schedule 13G	Panel B: Fi	ling date – 'clean	events Control sample			Difference	
	t = Filing date	(l)	(2)	(3)	(t)	(2)	(3)	(t)	(2)	(3)
CAAR	(-10, 0)	-0.005	-0.003	-0.004	-0.009*	-0.010*	-0.010*	0.004	0.007	0.006
	Pos: Neg	265:330	276:319	282:313	271:324	265:330	266:329			
	(-5, 0)	-0.001	0.000	0.000	-0.003	-0.003	0.003	0.001	0.004	0.003
	Pos: Neg	286:309	283:312	283:312	275:320	271:324	274:321			
	(-1, 0)	-0.003	-0.002	-0.002	-0.003*	-0.003*	-0.003	0.000	0.001	0.001
	Pos: Neg	281:314	288:307	292:301	256:339	258:337	266:329			
SCAAR	(-10, 0)	-0.024	-0.019	-0.020	-0.019*	-0.021*	-0.022*	-0.005	0.002	0.002
	(-5, 0)	-0.007	-0.004	-0.003	-0.006	-0.007	-0.007	-0.001	0.003	0.004
	(-1, 0)	-0.001	-0.001	0.001	-0.006	-0.007	-0.005	0.004	0.006	0.006
AVAR	(-10, 0)	2.809***	2.951***	2.988***	1.260***	1.309***	1.341***	1.550***	1.643***	<i>I.647</i> ***
	(-5, 0)	3.179***	3.337***	3.381***	1.378***	1.427***	1.469***	1.802***	$1.910^{***}$	1.912***
	(-1, 0)	4.237***	4.524***	4.578***	1.369***	1.427***	1.471***	2.869***	3.097***	3.107***
CAAV	(-10, 0)	$0.094^{***}$			0.011			0.083***		
	(-5, 0)	$0.070^{***}$			0.00			0.061***		
	(-1, 0)	$0.038^{***}$			0.006			0.032***		
Number o	f Obs.	595	595	595	595	595	595			
Notes: The the 1 resu	table presents CA <sup>A</sup> narket model (1), ti lts for the full samp	ARs, SCAARs, A he Carhart mode ole. Panel B prese	VAR, and CAAV 1(2), and the FF5 i ents results for the	<sup>7</sup> for Schedule 13G model (3). In each 'clean' events exc	i, a control sample case, we calculate sluding the events	e, and their differ e the number of where the ten-da	rence, $(-10, -5, -1)$ positive and negat ay disclosure peric	<ol> <li>days before the tive events. Pane od is violated. *,</li> </ol>	e event date us A presents the **, *** indica	ng e
stati	stical significance :	at the 10%, 5% a	nd 1% level, respe	sctively.						

 Table 10
 Abnormal returns, return volatility and volume of Schedule 13G filings, a control sample and their difference (continued)

		(3)	-0.012	0.015	0.027*	-0.033	0.058**	***060	218***	324***	274***				618	30
	g Schedule 13D	(2)	-0.012	-0.015	-0.028**	-0.035*	-0.060***	$-0.094^{***}$ $-0$	1.182*** 1.	1.292*** 1.	1.240*** 1.				618	or to the filing date (Lag prior to the filing date (La ). We calculate the above 0%. 5% and 1% level.
	Γa	(1)	-0.015*	-0.018*	$-0.034^{**}$	$-0.046^{**}$	-0.065 ***	$-0.110^{***}$	1.123***	$I.220^{***}$	1.175***	-0.012	0.006	-0.005	618	dar window in the year pri lendar window in the year 1 (2), and the FF5 model (3 istical sionificance at the 1
Schedule 13D		(3)	$0.124^{***}$	$0.040^{***}$	0.156***	0.385***	$0.114^{***}$	0.477 * * *	3.624***	1.502***	2.563***				618	ilings and for the same calen ilings and for the same calen odel (1), the Carhart mode of * ** *** indicate start
Panel A:	Schedule 13D	(2)	$0.124^{***}$	$0.040^{***}$	$0.157^{***}$	$0.387^{***}$	$0.114^{***}$	$0.478^{***}$	3.625***	1.502***	2.564***				618	V for Schedule 13D fili A), and Schedule 13G f B), using the market me
		(1)	0.128***	0.043 * * *	0.163***	0.382***	0.115***	0.472***	3.373***	1.382***	2.377***	0.301***	$0.120^{***}$	0.412***	618	ARs, AVAR, and CAA er the filing date (Panel er the filing date (Panel CRSP values for the m
		t = Filing date	(-45, 0)	(0, +45)	(-45, +45)	(-45, 0)	(0, +45)	(-45, +45)	(-45, 0)	(0, +45)	(-45, +45)	(-45, 0)	(0, +45)	(-45, +45)		presents CAARs, SCA (45 days before and aft (45 days before and aft for all events that have
		I	CAAR			SCAAR			AVAR			CAAV			Number of obs.	Notes: The table Schedule) Schedule) measures

Table 11Prior year abnormal returns, return volatility and volume of Schedule 13D and<br/>Schedule 13G filings

Schedule 13G         Lag Schedule 13G	Schedule 13G         Lag Schedule 13G           Filting date         (1)         (2)         (3)         (1)         (2)         (3)           Filting date         (1)         (2)         (3)         (1)         (2)         (3)           (-45, 0)         0.016         0.015         0.015         -0.020**         -0.017**         -0.018**           (-45, 0)         0.016         0.015         0.025***         0.027***         0.016         -0.014         -0.009         -0.016**           -45, +45)         0.039**         0.039***         0.038***         -0.024**         -0.014**         -0.014**         -0.014***           -45, +45)         0.066         0.066         0.066         -0.014**         -0.024***         -0.014***         -0.024***           (0, +45)         0.067***         0.073***         0.071***         -0.033***         -0.024***         -0.034***           (1, 45)         0.066***         0.071***         -0.033***         -0.024***         -0.034***           (1, 45)         0.066***         1.333***         1.317****         1.317****           (1, 45)         1.334***         1.323***         1.201***         -0.034***         1.617****				Panel B	: Schedule 13G			
ifling date(1)(2)(3)(1)(2)(3) $45,0$ $0.016$ $0.015$ $-0.020^{**}$ $-0.017^{**}$ $-0.0$ $45,0$ $0.025^{***}$ $0.015$ $0.015$ $-0.020^{***}$ $-0.009$ $-0.0$ $5,+45$ $0.025^{***}$ $0.027^{***}$ $0.026^{***}$ $-0.014$ $-0.009$ $-0.009$ $45,0$ $0.028^{***}$ $0.033^{***}$ $0.033^{***}$ $-0.009$ $-0.009$ $-0.009$ $45,0$ $0.008$ $0.006$ $0.072^{***}$ $-0.041^{***}$ $-0.041^{***}$ $-0.025^{***}$ $-0.014^{***}$ $45,0$ $0.067^{***}$ $0.072^{***}$ $0.072^{***}$ $-0.041^{***}$ $-0.033^{***}$ $-0.033^{***}$ $-0.033^{***}$ $-0.033^{***}$ $-0.033^{***}$ $45,0$ $0.067^{***}$ $0.072^{***}$ $0.071^{***}$ $-0.041^{***}$ $-0.074^{***}$ <t< th=""><th>ifing date(1)(2)(3)(1)(2)(3)-45.0)0.0160.0150.015-0.020**-0.017**-0.018**-45.0)0.025***0.025***0.025***-0.014-0.009-0.0105, +45)0.039**0.039***0.025***-0.026**-0.014-0.029**45.0)0.0880.0660.006-0.047***-0.014**-0.029**5, +45)0.0880.0660.006-0.047***-0.041**-0.034**5, +45)0.067**0.072***0.072***-0.041**-0.034**-0.034**5, +45)0.067**0.072***0.071***-0.041**-0.034**-0.034**45, 0)0.067**0.072***0.071**-0.041**-0.034**-0.034**45, 0)0.067**0.072***1.201***1.375***1.517***1.517***45, 0)0.067**0.071**0.071***0.067**-0.031***-0.031***45, 0)0.067**0.071**1.375***1.517***1.517***1.517***45, 0)0.06**1.334***1.327***1.335***1.517***1.517***45, 0)0.06***1.334***1.335***1.517***1.517***1.517***45, 0)0.06***1.334***1.335***1.517***1.517***1.517***45, 0)0.06***1.327***1.335***1.517***1.517***1.517***45, 0)0.06***1.334***1.337***1.535***1</th><th></th><th></th><th></th><th>Schedule 13G</th><th></th><th></th><th>Lag Schedule 13G</th><th></th></t<>	ifing date(1)(2)(3)(1)(2)(3)-45.0)0.0160.0150.015-0.020**-0.017**-0.018**-45.0)0.025***0.025***0.025***-0.014-0.009-0.0105, +45)0.039**0.039***0.025***-0.026**-0.014-0.029**45.0)0.0880.0660.006-0.047***-0.014**-0.029**5, +45)0.0880.0660.006-0.047***-0.041**-0.034**5, +45)0.067**0.072***0.072***-0.041**-0.034**-0.034**5, +45)0.067**0.072***0.071***-0.041**-0.034**-0.034**45, 0)0.067**0.072***0.071**-0.041**-0.034**-0.034**45, 0)0.067**0.072***1.201***1.375***1.517***1.517***45, 0)0.067**0.071**0.071***0.067**-0.031***-0.031***45, 0)0.067**0.071**1.375***1.517***1.517***1.517***45, 0)0.06**1.334***1.327***1.335***1.517***1.517***45, 0)0.06***1.334***1.335***1.517***1.517***1.517***45, 0)0.06***1.334***1.335***1.517***1.517***1.517***45, 0)0.06***1.327***1.335***1.517***1.517***1.517***45, 0)0.06***1.334***1.337***1.535***1				Schedule 13G			Lag Schedule 13G	
45,0 $0.016$ $0.015$ $0.015$ $-0.020**$ $-0.017**$ $-0.017**$ $-0.017**$ $-0.017**$ $-0.017**$ $-0.017**$ $-0.017***$ $-0.0125***$ $-0.0125***$ $-0.014$ $-0.009$ $-0.014$ $-0.009$ $-0.014$ $-0.009$ $-0.014$ $-0.009$ $-0.014$ $-0.009$ $-0.014$ $-0.009$ $-0.014$ $-0.009$ $-0.014$ $-0.009$ $-0.014$ $-0.009$ $-0.014$ $-0.009$ $-0.014$ $-0.009$ $-0.014$ $-0.0126**$ $-0.014$ $-0.0126**$ $-0.014$ $-0.0126**$ $-0.014$ $-0.0133***$ $-0.014$ $-0.0133***$ $-0.014$ $-0.0133***$ $-0.014$ $-0.0133***$ $-0.0133***$ $-0.0133****$ $-0.0133**********************************$	45,0) $0.016$ $0.015$ $0.015$ $0.015$ $-0.020^{**}$ $-0.017^{**}$ $-0.017^{**}$ $-0.017^{**}$ $-0.017^{**}$ $-0.010^{**}$ 5,45) $0.028^{***}$ $0.028^{***}$ $0.028^{***}$ $-0.014$ $-0.009$ $-0.010^{***}$ 45,0) $0.038^{***}$ $0.026^{***}$ $0.026^{***}$ $-0.024^{***}$ $-0.041^{***}$ $-0.028^{***}$ 45,0) $0.008$ $0.006$ $0.006$ $-0.047^{***}$ $-0.024^{***}$ $-0.041^{***}$ 45,0) $0.067^{***}$ $0.07^{***}$ $0.07^{***}$ $-0.041^{***}$ $-0.041^{***}$ 45,0) $1.858^{***}$ $0.07^{***}$ $0.07^{***}$ $-0.041^{***}$ $-0.033^{***}$ 45,0) $1.858^{***}$ $2.002^{***}$ $0.071^{***}$ $-0.041^{***}$ $-0.031^{***}$ 45,0) $1.858^{***}$ $1.455^{***}$ $1.457^{***}$ $1.457^{***}$ $1.457^{***}$ $1.468^{***}$ 5,45) $1.60^{***}$ $1.73^{***}$ $1.492^{***}$ $1.221^{***}$ $1.468^{***}$ $1.468^{***}$ 5,45) $1.60^{***}$ $1.73^{***}$ $1.290^{***}$ $1.468^{***}$ $1.468^{***}$ 45,0) $0.66^{***}$ $1.73^{***}$ $1.221^{***}$ $1.468^{***}$ $1.60^{***}$ 45,0) $0.067^{**}$ $0.07^{**}$ $1.33^{***}$ $1.468^{***}$ $1.468^{***}$ 45,0) $0.66^{***}$ $1.73^{***}$ $1.822^{***}$ $1.468^{***}$ $1.468^{***}$ 45,0) $0.66^{***}$ $1.33^{***}$ $1.230^{***}$ $1.468^{***}$ $1.468^{***}$ 47,0)	I = I	iling date	(t)	(2)	(3)	(t)	(2)	(3)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	0, +45) $0,025**$ $0,027***$ $0,026**$ $-0.009$ $-0.009$ $-0.010$ $45, +45$ ) $0,039**$ $0,038**$ $0,038**$ $-0,033**$ $-0.026**$ $-0.029**$ $45, 0$ ) $0,008$ $0,006$ $0,006$ $-0,041**$ $-0.024**$ $-0.024**$ $45, 0$ ) $0,008$ $0,006$ $0,006$ $-0,041**$ $-0.041**$ $-0.026**$ $45, 0$ ) $0,062**$ $0,072***$ $0,072***$ $-0.041**$ $-0.024**$ $-0.026**$ $45, +45$ ) $0,067**$ $0,072***$ $0,071**$ $-0,041**$ $-0.034**$ $-0.034**$ $45, +45$ ) $1,838***$ $1,452***$ $1,221***$ $1,230***$ $1,317***$ $45, +45$ ) $1,334***$ $1,492***$ $1,221***$ $1,230***$ $1,617***$ $45, +45$ ) $1,334***$ $1,733***$ $1,327***$ $1,468***$ $1,467**$ $1,337***$ $45, +45$ ) $0,066***$ $1,733***$ $1,323***$ $1,468***$ $1,467**$ $1,467**$ $45, +45$ ) $0,066***$ $1,733***$ $1,323***$ $1,468***$ $1,467**$ $1,467**$ $45, +45$ ) $0,066***$ $1,733***$ $1,323***$ $1,468***$ $1,467**$ $1,468***$ $45, +45$ ) $0,066***$ $1,733***$ $1,323***$ $1,468***$ $1,467**$ $1,468***$ $45, +45$ ) $0,066***$ $1,733***$ $1,323***$ $1,468***$ $1,328***$ $1,468***$ $45, +45$ ) $0,066***$ $1,733***$ $1,323****$ $1,323****$ $1,468***$	$\sim$	45, 0)	0.016	0.015	0.015	-0.020 **	$-0.017^{**}$	$-0.018^{**}$
$45, +45$ $0.039^{**}$ $0.039^{***}$ $0.038^{***}$ $-0.03^{***}$ $-0.026^{***}$ $-0.0$ $45, 0$ $0.008$ $0.006$ $0.041^{***}$ $-0.041^{***}$ $-0.041^{***}$ $-0.041^{***}$ $-0.01^{***}$	45, +45) $0.039^{**}$ $0.039^{***}$ $0.033^{***}$ $-0.026^{***}$ $-0.029^{***}$ 45, 0) $0.008$ $0.006$ $0.006$ $-0.041^{***}$ $-0.041^{***}$ $-0.044^{***}$ 45, 10) $0.062^{***}$ $0.005^{***}$ $0.072^{***}$ $0.024^{***}$ $-0.041^{***}$ $-0.041^{***}$ 45, +45) $0.062^{***}$ $0.072^{***}$ $0.071^{***}$ $-0.041^{***}$ $-0.041^{***}$ $-0.044^{***}$ 45, 10) $0.067^{***}$ $0.073^{***}$ $0.071^{***}$ $1.204^{***}$ $-0.033^{**}$ $-0.034^{***}$ 45, 10) $1.85^{***}$ $1.492^{***}$ $1.221^{***}$ $1.290^{***}$ $1.617^{***}$ 45, 10) $1.85^{***}$ $1.733^{***}$ $1.492^{***}$ $1.335^{***}$ $1.617^{***}$ 45, 10) $0.06^{***}$ $1.733^{***}$ $1.802^{***}$ $1.335^{***}$ $1.607^{***}$ $1.617^{***}$ 45, 10) $0.06^{***}$ $0.016$ $0.016$ $0.016$ $1.468^{***}$ $1.617^{***}$ 45, 445) $0.066^{***}$ $1.332^{***}$ $1.335^{***}$ $1.335^{***}$ $1.608^{***}$ $0.143$ $0.016$ $0.016$ $0.016$ $0.016$ $0.016$ $0.453$ $0.066^{***}$ $0.006^{***}$ $0.0016$ $0.016$ $0.453$ $0.016$ $0.025$ $0.041$ $930$ $0.230^{***}$ $0.230^{***}$ $0.041$ $930$ $930$ $0.245$ $0.230^{***}$ $0.230^{***}$ $0.041$ $930$ $0.230^{***}$ $0.230^{***}$ $0.041$ $930$ $930$ <	e	0, +45)	0.025***	0.027***	0.026***	-0.014	-0.009	-0.010
45,0 $0.008$ $0.006$ $-0.047***$ $-0.041**$ $-0.01**$ $-0.01**$ $0,+45$ $0.062***$ $0.072***$ $0.072***$ $-0.041**$ $-0.041**$ $-0.033*$ $-0.01$ $45,+45$ $0.062***$ $0.072***$ $0.072***$ $-0.041**$ $-0.033*$ $-0.01$ $45,+45$ $0.067**$ $0.072***$ $0.071**$ $-0.037***$ $-0.027****$ $-0.027*****$ $-0.027*****$ $-0.027*****$ $-0.027****$	45, 0 $0.008$ $0.006$ $0.047**$ $-0.047**$ $-0.041**$ $-0.041**$ $-0.044**$ $0, +45$ $0.062***$ $0.072***$ $0.072***$ $-0.041**$ $-0.033*$ $-0.034**$ $45, +45$ $0.067**$ $0.072***$ $0.071**$ $-0.041**$ $-0.031**$ $-0.041***$ $45, +45$ $0.067**$ $0.071**$ $-0.041**$ $-0.031**$ $-0.034**$ $-0.034**$ $45, 0$ $1.858**$ $2.002***$ $2.085***$ $1.221***$ $1.290***$ $1.317***$ $5, +45$ $1.334**$ $1.492***$ $1.422***$ $1.290***$ $1.617***$ $5, +45$ $1.334**$ $1.492***$ $1.231***$ $1.617***$ $45, 0$ $0.06***$ $1.733***$ $1.492***$ $1.487***$ $1.617***$ $45, 0$ $0.160***$ $1.335***$ $1.468***$ $0.016$ $1.487**$ $1.468***$ $45, 0$ $0.160***$ $1.335***$ $1.487***$ $1.468***$ $1.468***$ $45, 0$ $0.06***$ $1.733***$ $1.362***$ $1.468***$ $1.468***$ $45, 0$ $0.06***$ $0.06***$ $0.016$ $0.025$ $1.468***$ $5, +45$ $0.230**$ $0.025$ $0.025$ $0.025$ $45, +45$ $0.230**$ $0.021$ $0.025$ $0.021$ $5, +45$ $0.230**$ $0.021$ $0.021$ $0.021$ $5, +45$ $0.230**$ $0.021$ $0.021$ $0.041$ $500$ $930$ $930$ $930$ $930$ $500$ $0.06***$ $0.041$	Ţ	45, +45)	$0.039^{**}$	$0.039^{***}$	$0.038^{***}$	-0.033***	$-0.026^{**}$	-0.029 **
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	, +45 $0.062**$ $0.072***$ $0.072***$ $-0.041**$ $-0.033*$ $-0.033*$ $-0.036*$ $45, +45$ $0.067**$ $0.073**$ $0.071**$ $-0.087***$ $-0.037***$ $-0.031***$ $45, 0$ $1.858**$ $2.002***$ $2.002***$ $1.221***$ $1.200***$ $1.317***$ $45, 0$ $1.858**$ $2.002***$ $1.492***$ $1.487***$ $1.583**$ $1.617***$ $45, 0$ $0.168**$ $1.733***$ $1.802***$ $1.607***$ $1.617***$ $45, 0$ $0.168***$ $1.733***$ $1.334**$ $1.617***$ $45, 0$ $0.168***$ $1.733***$ $1.335***$ $1.617***$ $45, 0$ $0.168**$ $1.733***$ $1.335***$ $1.617***$ $45, 0$ $0.168**$ $0.016$ $1.487**$ $1.687**$ $45, 0$ $0.168**$ $0.016$ $0.16$ $1.468**$ $0.168**$ $0.006$ $0.006$ $0.016$ $0.016$ $5, +45$ $0.230**$ $0.006$ $0.025$ $0.041$ $930$ $930$ $930$ $930$ $930$ $930$ $930$ $930$ $930$ $920$ $930$ $930$ $930$ $920$ $930$ $930$ $930$ $8$ before and after the filing date (Panel B), using the market model (1), the Carhart model (2), and the FF5 model (3). We calculate the above $8$ before and after the filing date (Panel B), using the market model (1), the Carhart model (2), and the FF5 model (9) evel $8$ before and after the filing date (Panel B), using the market model (1), the Carhart mo	÷	-45, 0)	0.008	0.006	0.006	-0.047 * * *	-0.041 **	$-0.044^{**}$
$15, +45$ $0.07^{**}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.07^{***}$ $0.01^{***}$ $1.29^{***}$ $1.31^{***}$ $1.29^{***}$ $1.31^{***}$ $1.29^{***}$ $1.31^{****}$ $1.31^{****}$ $1.31^{****}$ $1.33^{****}$ $1.33^{****}$ $1.43^{****}$ $1.43^{****}$ $1.43^{****}$ $1.43^{****}$ $1.43^{****}$ $1.43^{****}$ $1.43^{****}$ $1.43^{****}$ $1.43^{****}$ $1.43^{****}$ $1.43^{****}$ </td <td>15, +45<math>0.067**</math><math>0.073**</math><math>0.071**</math><math>-0.087**</math><math>-0.074**</math><math>-0.081**</math><math>45, 0</math><math>1.858**</math><math>2.002***</math><math>2.002***</math><math>2.085***</math><math>1.221***</math><math>1.290***</math><math>1.317***</math><math>1, +45</math><math>1.334**</math><math>1.455***</math><math>1.455***</math><math>1.221***</math><math>1.290***</math><math>1.617***</math><math>5, +45</math><math>1.334**</math><math>1.455***</math><math>1.455***</math><math>1.617***</math><math>1.617***</math><math>45, 0</math><math>0.06**</math><math>1.733***</math><math>1.802***</math><math>1.687***</math><math>1.617***</math><math>45, 0</math><math>0.168**</math><math>1.733***</math><math>1.607**</math><math>1.468**</math><math>1.667**</math><math>45, 0</math><math>0.168**</math><math>1.733***</math><math>1.355***</math><math>1.468**</math><math>1.667**</math><math>5, +45</math><math>0.066**</math><math>0.066**</math><math>0.016</math><math>0.016</math><math>0.016</math><math>5, +45</math><math>0.230**</math><math>0.025</math><math>0.041</math><math>9.025</math><math>5, +45</math><math>0.230**</math><math>0.021</math><math>0.041</math><math>9.02</math><math>5, -45</math><math>0.230**</math><math>0.041</math><math>9.00</math><math>9.00</math><math>9.30</math><math>9.30</math><math>9.30</math><math>9.30</math><math>9.30</math><math>5 e 6 6 7 e and a fler the film g date (Panel A), and Shedule 130 film s and for the same calendar window in the year prior to the film g date (Lag s before and after the film g date (Panel B), using the market model (1), the Carhart model (2), and the FF5 model (3). We calculate the above vertice that have CRSP values for the prior year estimation period. *, *, **** indicate statistical significance at the 10%, 5% and 1% evel.</math></td> <td>Ξ</td> <td>), +45)</td> <td>0.062***</td> <td>0.072***</td> <td>0.072***</td> <td>-0.041 **</td> <td>-0.033*</td> <td>-0.036*</td>	15, +45 $0.067**$ $0.073**$ $0.071**$ $-0.087**$ $-0.074**$ $-0.081**$ $45, 0$ $1.858**$ $2.002***$ $2.002***$ $2.085***$ $1.221***$ $1.290***$ $1.317***$ $1, +45$ $1.334**$ $1.455***$ $1.455***$ $1.221***$ $1.290***$ $1.617***$ $5, +45$ $1.334**$ $1.455***$ $1.455***$ $1.617***$ $1.617***$ $45, 0$ $0.06**$ $1.733***$ $1.802***$ $1.687***$ $1.617***$ $45, 0$ $0.168**$ $1.733***$ $1.607**$ $1.468**$ $1.667**$ $45, 0$ $0.168**$ $1.733***$ $1.355***$ $1.468**$ $1.667**$ $5, +45$ $0.066**$ $0.066**$ $0.016$ $0.016$ $0.016$ $5, +45$ $0.230**$ $0.025$ $0.041$ $9.025$ $5, +45$ $0.230**$ $0.021$ $0.041$ $9.02$ $5, -45$ $0.230**$ $0.041$ $9.00$ $9.00$ $9.30$ $9.30$ $9.30$ $9.30$ $9.30$ $5 e 6 6 7 e and a fler the film g date (Panel A), and Shedule 130 film s and for the same calendar window in the year prior to the film g date (Lag s before and after the film g date (Panel B), using the market model (1), the Carhart model (2), and the FF5 model (3). We calculate the above vertice that have CRSP values for the prior year estimation period. *, *, **** indicate statistical significance at the 10%, 5% and 1% evel.$	Ξ	), +45)	0.062***	0.072***	0.072***	-0.041 **	-0.033*	-0.036*
45,0 $1.858**$ $2.002***$ $2.085**$ $1.221***$ $1.290***$ $1.3$ $0,+45$ $1.334**$ $1.455***$ $1.492***$ $1.487***$ $1.585***$ $1.61$ $15,+45$ $1.334**$ $1.733***$ $1.492***$ $1.487***$ $1.438***$ $1.61$ $45,0$ $0.168***$ $0.016$ $0.016$ $0.016$ $0.025$	45.0) $1.836**$ $2.002***$ $2.085***$ $1.221***$ $1.290***$ $1.317***$ $0.+45$ ) $1.334**$ $1.455***$ $1.450***$ $1.317***$ $1.317***$ $1.317***$ $15.+45$ ) $1.334**$ $1.455***$ $1.458***$ $1.607***$ $1.617***$ $45.0$ ) $0.160***$ $1.733***$ $1.602***$ $1.468***$ $1.468***$ $45.0$ ) $0.168***$ $0.016$ $1.438***$ $1.468***$ $45.0$ ) $0.166***$ $0.166$ $1.468***$ $0.166$ $1.+45$ $0.066***$ $0.006$ $0.016$ $0.230***$ $0.006$ $0.001$ $0.001$ $0.230***$ $0.0041$ $0.025$ $0.0041$ $0.230***$ $0.0041$ $0.025$ $0.0041$ $0.230***$ $0.0041$ $0.025$ $0.0041$ $0.230***$ $0.025$ $0.0041$ $0.025$ $0.230***$ $0.0041$ $0.025$ $0.0041$ $0.230***$ $0.0041$ $0.025$ $0.0041$ $0.230***$ $0.0041$ $0.025$ $0.0041$ $0.230***$ $0.0041$ $0.025$ $0.0041$ $0.230***$ $0.0041$ $0.025$ $0.0041$ $0.230***$ $0.0041$ $0.025$ $0.0041$ $0.230***$ $0.0041$ $0.025$ $0.0041$ $0.230***$ $0.0041$ $0.025$ $0.0041$ $0.230***$ $0.0041$ $0.0041$ $0.0041$ $0.230***$ $0.0041$ $0.0041$ $0.0041$ $0.016***********************************$	Ĵ	15, +45)	$0.067^{**}$	0.073**	0.071 **	$-0.087^{***}$	$-0.074^{***}$	$-0.081^{***}$
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$(, +45)$ $[.334^{***}]$ $[.457^{***}]$ $[.467^{***}]$ $[.367^{***}]$ $[.517^{***}]$ $(45, -45)$ $1.600^{***}$ $[.733^{***}]$ $[.302^{***}]$ $[.438^{***}]$ $[.468^{***}]$ $(45, 0)$ $0.160^{***}$ $[.733^{***}]$ $[.355^{***}]$ $[.438^{***}]$ $[.468^{***}]$ $(45, 0)$ $0.168^{***}$ $[.733^{***}]$ $[.355^{***}]$ $[.438^{***}]$ $[.468^{***}]$ $(45, 0)$ $0.166^{***}$ $0.016^{***}$ $[.468^{***}]$ $[.468^{***}]$ $(5, +45)$ $0.230^{***}]$ $0.0016^{***}$ $[.468^{***}]$ $(5, +45)$ $0.230^{***}]$ $0.025^{***}$ $[.401]^{***}$ $(5, +45)$ $0.230^{***}]$ $0.00116^{***}$ $[.468^{***}]$ $(5, +45)$ $0.230^{***}$ $0.00116^{***}$ $[.468^{***}]$ $(5, +45)$ $0.230^{***}$ $0.00116^{***}$ $[.468^{***}]$ $(5, +45)$ $0.230^{***}$ $0.00116^{***}$ $[.468^{***}]$ $(5, +45)$ $0.230^{***}$ $0.00116^{***}$ $[.468^{***}]$ $(5, +45)$ $0.230^{***}$ $0.00116^{***}$ $[.468^{***}]$ $(5, +45)$ $0.230^{***}$ $0.00116^{***}$ $[.468^{***}]$ $(5, +45)$ $0.230^{***}$ $0.00116^{***}$ $[.468^{***}]$ $(5, +45)$ $0.230^{***}$ $0.0011^{***}$ $0.00116^{****}$ $(5, +45)$ $0.230^{***}$ $0.0011^{****}$ $0.00116^{****}$ $(5, +45)$ $0.230^{***}$ $0.00116^{****}$ $0.00116^{*****}$ $(5, +45)$ $0.230^{****}$ $0.0011^{**************************$	÷	-45, 0)	$I.858^{***}$	2.002***	2.085***	1.221***	$I.290^{***}$	1.317***
45, +45       1.600***       1.733***       1.802***       1.355***       1.438***         -45, 0)       0.168***       0.016       0.016         .+45)       0.066***       0.025	$15, 45$ $1.600^{***}$ $1.733^{***}$ $1.802^{***}$ $1.355^{***}$ $1.438^{***}$ $1.468^{***}$ $45, 0$ $0.168^{***}$ $0.016$ $0.016$ $0.016$ $1, +45$ $0.066^{***}$ $0.016$ $0.025$ $15, +45$ $0.026$ $0.025$ $0.021$ $15, +45$ $0.230^{***}$ $0.041$ $930$	Ξ	), +45)	1.334***	1.455***	1.492***	1.487***	1.585***	1.617***
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Ĵ	15, +45)	$I.600^{***}$	1.733***	1.802***	1.355***	1.438***	$I.468^{***}$
), +45) 0, 0/66*** 0.025	0.45)       0.066***       0.025         45, +45)       0.230***       0.041         45, +45)       0.230***       0.041         930       930       930       930         930       930       930       930       930         s CAARs, SCAARs, AVAR, and CAAV for Schedule 13D filings and for the same calendar window in the year prior to the filing date (Lag s before and after the filing date (Panel B), using the market model (1), the Carhart model (2), and the FF5 model (3). We calculate the above vents that have CRSP values for the prior year estimation period. *, ***, *** indicate statistical significance at the 10%, 5% and 1% level,	÷	-45, 0)	0.168***			0.016		
	45, +45)       0.230***       0.041         930       930       930       930       930         936       930       930       930       930       930         scAARs, SCAARs, AVAR, and CAAV for Schedule 13D filings and for the same calendar window in the year prior to the filing date (Lag s before and after the filing date (Panel A), and Schedule 13G filings and for the same calendar window in the year prior to the filing date (Lag s before and after the filing date (Panel B), using the market model (1), the Carhart model (2), and the FF5 model (3). We calculate the above vents that have CRSP values for the prior year estimation period. *, ***, *** indicate statistical significance at the 10%, 5% and 1% level,	Ξ	), +45)	$0.066^{***}$			0.025		
5, +45) 0.230*** 0.041	930930930930930930930930s CAARs, SCAARs, AVAR, and CAAV for Schedule 13D filings and for the same calendar window in the year prior to the filing date (Lag s before and after the filing date (Panel A), and Schedule 13G filings and for the same calendar window in the year prior to the filing date (Lag s before and after the filing date (Panel A), using the market model (1), the Carhart model (2), and the FF5 model (3). We calculate the above vents that have CRSP values for the prior year estimation period. *, ***, **** indicate statistical significance at the 10%, 5% and 1% level,	Ţ	15, +45)	$0.230^{***}$			0.041		
930 930 930 930 930 930 9	s CAARs, SCAARs, AVAR, and CAAV for Schedule 13D filings and for the same calendar window in the year prior to the filing date (Lag s before and after the filing date (Panel A), and Schedule 13G filings and for the same calendar window in the year prior to the filing date (Lag s before and after the filing date (Panel B), using the market model (1), the Carhart model (2), and the FF5 model (3). We calculate the above such that have CRSP values for the prior year estimation period. *, **, *** indicate statistical significance at the 10%, 5% and 1% level,			930	930	930	930	930	930

Table 11Prior year abnormal returns, return volatility and volume of Schedule 13D and<br/>Schedule 13G filings (continued)

			Schedule 13D	Panel A:	Filing date – all	events Schedule 13G			Difference	
I	t = Filing date	<i>(t)</i>	(2)	(3)	(v)	(2)	(3)	(t)	(2)	(3)
CAAR	(-45, 0)	0.126***	0.128***	$0.130^{***}$	-0.016	-0.013	-0.013	0.142***	$0.141^{***}$	0.143***
	(0, +45)	$0.046^{***}$	$0.048^{***}$	$0.047^{***}$	0.028***	0.035***	$0.034^{***}$	$0.024^{***}$	$0.013^{***}$	0.013***
	(-45, +45)	0.163***	0.168***	0.168***	0.010	$0.018^{***}$	$0.018^{***}$	0.153***	$0.150^{***}$	$0.150^{***}$
SCAAR	(-45, 0)	0.383***	$0.407^{***}$	$0.415^{***}$	-0.057*	-0.050*	-0.048	$0.440^{***}$	$0.157^{***}$	0.463***
	(0, +45)	0.116***	0.130***	0.130***	$0.080^{***}$	0.105***	0.102***	0.036***	0.025***	$0.028^{***}$
	(-45, +45)	$0.470^{***}$	0.508***	$0.516^{***}$	0.020	0.049	0.046	0.450***	0.459***	$0.470^{***}$
AVAR	(-45, 0)	3.757***	$4.050^{***}$	4.151***	1.550***	<i>1.647</i> ***	<i>I.687</i> ***	2.207***	2.403***	2.464***
	(0, +45)	1.377***	$I.506^{***}$	1.544***	1.331***	<i>I.467</i> ***	1.499***	$0.046^{***}$	$0.039^{***}$	0.045***
	(-45, +45)	2.567***	2.778***	$2.850^{***}$	<i>I.446***</i>	1.563***	1.560***	1.121***	1.215***	1.290***
CAAV	(-45, 0)	0.345***			0.136***			0.209***		
	(0, +45)	0.146***			$0.048^{***}$			0.098***		
	(-45, +45)	0.481***			0.182***			0.299***		
Number of ol	<b>JS.</b>	468	468	468	468	468	468			
Notes: The tab after the coarsen nearest change events e	le presents CAARs) e filing date using th ed exact matching n neighbour. The coar in the target firm RC xcluding the events	SCAARs, AV/ e market model nethod and matc sened exact mai DA measured be where the ten-d	AR, and CAAV (1), the Carhart shing each event tching is estimat stween years t – lav disclosure po	for Schedule 131 t model (2), and 1 t firm to a Schedi ted using (log) fi ted using $(\log)$ fi and $t - 1$ . Panel 2 and i s violated.	D, a control samp the FF5 model (3) ule 13D event fro rm size, market-tt 1 A presents the ru * ** indical	le of Schedule 1 The control se m the same yea o-book ratio, rei seults for the fu te statistical sig	3G filings, and the maple of Schedule of Schedule r and the same incluration assets (RC III sample. Panel E III sample. Panel E Inificance at the 10 militicance at the 10 milit	neir difference, 42 e 13G filings is fc dustry (two-digit )A) measured at 3 presents results 10%, 5% and 1% 1	5 days before ar prmed by using SIC) with the t-1, and the for the 'clean' evel. respective	ithe slv.

Table 12Abnormal returns, return volatility and volume of Schedule 13D filings, a control<br/>sample of Schedule 13G filings and their difference

				Panel A:	Filing date – all	events				
		Schedu	de 13D			Schedule 13G			Difference	
I	t = Filing date	(v)	(2)	(3)	(1)	(2)	(3)	(t)	(2)	(3)
CAAR	(-45, 0)	0.122***	0.121***	0.123***	-0.006	-0.003	-0.005	0.128***	$0.124^{***}$	0.128***
	(0, +45)	$0.064^{***}$	0.066***	$0.064^{***}$	0.031**	0.038***	0.036***	0.033 * * *	$0.028^{***}$	0.028***
	(-45, +45)	0.179***	$0.180^{***}$	$0.180^{***}$	0.025	$0.034^{*}$	0.029	$0.154^{***}$	$0.146^{***}$	0.161***
SCAAR	(-45, 0)	0.325***	$0.343^{***}$	$0.347^{***}$	-0.044	-0.031	-0.036	0.369***	$0.374^{***}$	0.383***
	(0, +45)	0.165***	$0.174^{***}$	$0.174^{***}$	0.091***	0.116***	$0.109^{***}$	0.075***	$0.026^{***}$	0.065***
	(-45, +45)	0.469***	0.497***	0.501 ***	0.046	0.082	0.069	0.423***	$0.415^{***}$	0.432***
AVAR	(-45, 0)	3.700***	3.972***	$4.064^{***}$	1.585***	$1.670^{***}$	1.713***	2.115***	2.302***	2.351***
	(0, +45)	I.789***	1.966***	$2.002^{***}$	1.312***	<i>I.468***</i>	1.502***	0.477***	$0.498^{***}$	0.500***
	(-45, +45)	2.751***	2.978***	3.041***	<i>I.453***</i>	1.574***	1.612***	1.298***	$1.404^{***}$	1.429***
CAAV	(-45, 0)	$0.294^{***}$			0.115***			0.179***		
	(0, +45)	0.151***			0.036			0.115***		
	(-45, +45)	$0.438^{***}$			0.149**			0.289***		
Number of ol	<b>3</b> S.	273	273	273	273	273	273			
Notes: The tab after thu coarsen nearest change events e	le presents CAARs) e filing date using th ed exact matching n neighbour. The coar in the target firm RC xcluding the events	SCAARs, AVA e market model nethod and matc sened exact mat )A measured be where the ten-d	AR, and CAAV (1), the Carhart hing each event iching is estimate tween years $t \rightarrow \frac{1}{2}$ av disclosure p	for Schedule 13I i model (2), and t i firm to a Schedu ted using (log) fin and t - 1. Panel sriod is violated.	D, a control samp the FF5 model (3) ule 13D event fro rrm size, market-tr 1 A presents the ru * ** *** indica	le of Schedule J . The control sc m the same yea o-book ratio, re scults for the fu te statistical sig	3G filings, and th umple of Schedul r and the same in turn on assets (R( Il sample. Panel I mificance at the 1	heir difference, 42 e 13G filings is fc dustry (two-digit OA) measured at 3 presents results 0%, 5% and 1% 1	5 days before an ormed by using SIC) with the $t-1$ , and the for the 'clean' evel. respective	id the the

Table 12Abnormal returns, return volatility and volume of Schedule 13D filings, a control<br/>sample of Schedule 13G filings and their difference (continued)

			Schedule 13D	Panel.	A: Event date – $\alpha$	all events Schedule 13G			Difference	
·	t = Filing date	(1)	(2)	(3)	(1)	(2)	(3)	(1)	(2)	(3)
CAAR	(-10, 0)	0.051***	0.051***	0.052***	-0.010	-0.007	-0.007	0.061***	$0.058^{***}$	0.059***
	(-5, 0)	0.042***	0.043 * * *	0.043***	-0.004	-0.002	-0.001	0.038***	0.045***	$0.044^{***}$
	(-1, 0)	0.017***	$0.018^{***}$	0.019***	-0.000	0.000	0.001	$0.017^{***}$	$0.018^{***}$	$0.018^{***}$
SCAAR	(-10, 0)	0.149***	0.156***	0.159***	-0.032*	-0.026	-0.027	0.181***	0.182***	0.186***
	(-5, 0)	0.121***	0.128***	0.130***	-0.014	-0.009	-0.007	0.135***	0.137***	$0.137^{***}$
	(-1, 0)	$0.049^{***}$	0.053***	0.053***	-0.003	-0.002	-0.000	0.052***	0.055***	$0.053^{***}$
AVAR	(-10, 0)	6.639***	7.134***	7.270***	2.343***	2.513***	2.569***	4.296***	4.621***	4.701***
	(-5, 0)	8.448***	9.095***	9.252***	2.922***	3.110***	3.170***	5.526***	5.985***	6.082***
	(-1, 0)	12.129***	12.842 ***	13.027***	3.570***	3.864***	3.945***	8.559***	8.978***	9.082***
CAAV	(-10, 0)	0.158***			0.066***			0.092***		
	(-5, 0)	0.105***			$0.048^{***}$			0.057***		
	(-1, 0)	$0.054^{***}$			0.026***			0.028***		
Number o	f obs.	468	468	468	468	468	468			
Notes: The befo	table presents the re the event date u	CAARs, SCAAl Ising the market	Rs, AVAR, and C model (1), the Ci	CAAV for Schedu arhart model (2) a	le 13D, a control nd the FF5 mode	l sample of Sche 1 (3). The contr	edule 13G filings, ol sample of Sche	and their differend dule 13G filings is	ce, (-10, -5, -1) s formed by usin	days ig the
coai near	sened exact match est neighbour. The	ting method and coarsened exact	matching each er t matching is esti	vent firm to a Sch imated using (log)	firm size, marke	rrom the same to the same to the second seco	year and the same return on assets (	Industry (two-digi ROA) measured at	It SIC) with the $t t - 1$ , and the	
char ever	ige in the target fir ts excluding the ev	m ROA measure vents where the t	ed between years ten-day disclosur	t t - 3 and $t - 1$ . Pa te period is violate	mel A presents th d. *, **, *** ind	he results for the icate statistical	e full sample. Pan significance at the	el B presents result 2 10%, 5% and 1%	ts for the 'clean' level, respectiv	ely.

Table 13Abnormal returns, return volatility and volume of Schedule 13D filings, a control<br/>sample of Schedule 13G filings and their difference

Schedule 13D         Schedule 13G         Difference           ing date         (1)         (2)         (3)         (1)         (2)         (3)           0,0 $0.054^{***}$ $0.057^{***}$ $0.057^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.054^{***}$ $0.022^{***}$ 0,0 $0.048^{***}$ $0.022^{***}$ $0.002$ $0.001$ $0.002$ $0.024^{***}$ $0.024^{***}$ $0.024^{***}$ $0.024^{***}$ $0.024^{***}$ $0.024^{***}$ $0.024^{***}$ $0.024^{***}$ $0.022^{***}$ 0,0 $0.143^{***}$ $0.141^{***}$ $0.0211$ $0.002$ $0.001$ $0.022^{***}$ $0.024^{***}$ $0.022^{***}$ $0.022^{***}$ $0.022^{***}$ $0.022^{***}$ $0.022^{***}$ $0.022^{***}$ $0.022^{***}$ $0.022^{***}$ $0.022^{***}$ $0.022^{***}$ $0.022^{***}$ $0.022^{****}$ $0.022^{****}$ $0$	$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$					Panel B:	<i>Event date – 'cl</i>	ean' events				
gdate(1)(2)(3)(1)(2)(3)(1)(2)(3)(1)(2)(3)(0) $0.054***$ $0.052***$ $0.022***$ $0.032$ $0.038***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.052***$ $0.022***$ $0.022***$ $0.022***$ $0.022***$ $0.022***$ $0.024***$ $0.024***$ $0.024***$ $0.024***$ $0.022**$	$q_{date}$ (1)(2)(3)(1)(2)(3)(1)(2)(1) $0.054***$ $0.052***$ $0.022$ $0.038***$ $0.054***$ $0.054***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.022****$ <td></td> <td></td> <td></td> <td>Schedule 13D</td> <td></td> <td></td> <td>Schedule 13C</td> <td>Ĺ</td> <td></td> <td>Difference</td> <td></td>				Schedule 13D			Schedule 13C	Ĺ		Difference	
(0) $0.054***$ $0.052***$ $0.052***$ $0.022***$ $0.032***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.054***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.048***$ $0.049***$ $0.049***$ $(0)$ $0.022***$ $0.022**$ $0.002$ $-0.000$ $0.022$ $0.048***$ $0.048***$ $0.048***$ $0.049***$ $0.049***$ $(0)$ $0.122***$ $0.124***$ $0.022***$ $-0.001$ $-0.000$ $0.022$ $0.048***$ $0.049***$ $0.023***$ $0.023***$ $0.023***$ $(0)$ $0.127***$ $0.123***$ $0.121***$ $-0.001$ $-0.001$ $-0.001$ $0.022**$ $0.022***$ $0.054***$ $0.054***$ $0.151***$ $0.153***$ $(0)$ $0.028***$ $0.060***$ $0.061***$ $-0.011$ $-0.010$ $-0.008$ $0.066***$ $0.161***$ $0.153***$ $0.153***$ $0.153***$ $0.153***$ $(0)$ $0.058***$ $10.669***$ $10.659***$ $2.011***$ $2.010***$ $2.164***$ $0.167***$ $0.050***$ $0.050***$ $(0)$ $0.058***$ $10.66***$ $10.66***$ $2.011***$ $2.164***$ $2.23***$ $1.570***$ $1.632***$ $(0)$ $0.022***$ $1.070***$ $2.164***$ $2.245***$ $2.164***$ $2.23***$ $1.050***$ $1.060***$ $(0)$ $0.022***$ <t< td=""><td><math>(0)</math><math>(0.54^{***})</math><math>(0.62^{***})</math><math>(0.62^{***})</math><math>(0.63^{***})</math><math>(0.63^{***})</math><math>(0.63^{***})</math><math>(0.63^{***})</math><math>(0.63^{***})</math><math>(0.63^{***})</math><math>(0.63^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.64^{***})</math><math>(0.67^{***})</math><math>(0.67^{***})</math><math>(0.67^{***})</math><math>(0.67^{***})</math><math>(0.67^{***})</math><math>(0.67^{***})</math><math>(0.74^{***})</math><math>(0.67^{***})</math><math>(0.67^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math><math>(0.74^{***})</math></td><td>-12</td><td>ng date</td><td>(1)</td><td>(2)</td><td>(3)</td><td>(1)</td><td>(2)</td><td>(3)</td><td>(t)</td><td>(2)</td><td>(3)</td></t<>	$(0)$ $(0.54^{***})$ $(0.62^{***})$ $(0.62^{***})$ $(0.63^{***})$ $(0.63^{***})$ $(0.63^{***})$ $(0.63^{***})$ $(0.63^{***})$ $(0.63^{***})$ $(0.63^{***})$ $(0.64^{***})$ $(0.67^{***})$ $(0.67^{***})$ $(0.67^{***})$ $(0.67^{***})$ $(0.67^{***})$ $(0.67^{***})$ $(0.74^{***})$ $(0.67^{***})$ $(0.67^{***})$ $(0.74^{***})$	-12	ng date	(1)	(2)	(3)	(1)	(2)	(3)	(t)	(2)	(3)
(0) $0.048**$ $0.047**$ $0.07**$ $-0.00$ $0.02$ $0.048**$ $0.022**$ $(0)$ $0.022**$ $0.022$ $-0.001$ $-0.001$ $-0.001$ $-0.001$ $0.024**$ $0.022**$ $0.022**$ $(0)$ $0.123**$ $0.143**$ $0.146**$ $0.128**$ $0.121**$ $0.128**$ $0.121**$ $0.123**$ $(0)$ $0.127**$ $0.128**$ $0.143**$ $-0.011$ $-0.001$ $-0.001$ $0.024**$ $0.161**$ $0.123**$ $(0)$ $0.028**$ $0.143**$ $0.143**$ $0.121**$ $-0.011$ $-0.001$ $-0.001$ $0.023**$ $0.153**$ $(0)$ $0.028**$ $0.164**$ $0.061**$ $0.011**$ $-0.011$ $-0.001$ $-0.001$ $0.023**$ $0.163**$ $0.161**$ $(0)$ $0.028**$ $0.060**$ $0.061**$ $0.011**$ $2.222**$ $2.362**$ $2.411**$ $4.233**$ $4.570**$ $4.520**$ $(0)$ $0.022**$ $1.0526**$ $10.659**$ $2.011**$ $2.012**$ $4.233**$ $4.570**$ $1.632**$ $(0)$ $0.022**$ $1.0526**$ $1.0.692**$ $2.111**$ $4.223**$ $4.570**$ $1.632**$ $(0)$ $0.029**$ $1.002**$ $0.020**$ $0.002***$ $0.002***$ $0.0$	$(0)$ $0.048^{***}$ $0.047^{***}$ $0.07^{***}$ $0.072^{***}$ $0.072^{***}$ $0.048^{****}$ $0.048^{****}$ $0.048^{****}$ $0.048^{****}$ $0.048^{****}$ $0.023^{****}$ $0.023^{****}$ $0.023^{****}$ $0.023^{****}$ $0.021^{****}$ $0.021^{*****}$ $0.021^{****}$ $0.021^{****}$ $0.021^{****}$ $0.021^{****}$ $0.021^{****}$ $0.021^{****}$ $0.021^{****}$ $0.018^{****}$ $0.161^{****}$ $0.1$ $(0)$ $0.127^{****}$ $0.123^{****}$ $0.143^{****}$ $0.00111$ $-0.002$ $0.004$ $0.138^{****}$ $0.161^{****}$ $0.1$ $(0)$ $0.127^{****}$ $0.128^{****}$ $0.0111$ $-0.002$ $-0.001$ $0.023^{****}$ $0.133^{****}$ $0.133^{****}$ $0.133^{****}$ $0.133^{****}$ $0.133^{****}$ $10.73^{****}$ $12.1^{*}$ $0.741^{****}$ $12.1^{*}$ $0.741^{****}$ $12.1^{*}$ $0.073^{****}$ $12.1^{*}$ $0.023^{****}$ $12.1^{*}$ $0.023^{****}$ $12.1^{*}$ $0.741^{****}$ $12.1^{*}$ $0.741^{****}$ $12.7^{*}$ $0.023^{****}$ $12.1^{*}$ $0.023^{****}$ $0.023^{****}$ $0.023^{****}$ <t< td=""><td>-</td><td>0, 0)</td><td><math>0.054^{***}</math></td><td>0.052***</td><td>0.052***</td><td>-0.004</td><td>-0.002</td><td>-0.003</td><td>0.058***</td><td><math>0.054^{***}</math></td><td>0.055***</td></t<>	-	0, 0)	$0.054^{***}$	0.052***	0.052***	-0.004	-0.002	-0.003	0.058***	$0.054^{***}$	0.055***
$0$ $0.02^{***}$	,0) $0.022^{***}$ $0.022^{***}$ $0.022^{***}$ $0.023^{****}$ $0.023^{****}$ $0.023^{****}$ $0.023^{****}$ $0.168^{****}$ $0.073^{****}$ $0.133^{****}$ $0.168^{****}$ $0.073^{****}$ $0.073^{****}$ $0.073^{****}$ $0.073^{****}$ $0.073^{****}$ $0.073^{****}$ $0.073^{****}$ $10.73^{****}$ $10.73^{****}$ $10.659^{****}$ $2.235^{****}$ $4.197^{****}$ $4.223^{****}$ $4.167^{****}$ $7.4$ 0,0 $0.058^{****}$ $10.659^{****}$ $2.931^{****}$ $4.197^{****}$ $4.223^{****}$ $4.257^{****}$ $12.10^{***}$ 0,0 $0.132^{****}$ $16.258^{***}$ $2.911^{****}$ $4.223^{***}$ $12.11^{***}$ $7.411^{***}$ $7.411^{***}$ $7.411^{***}$ $7.411^{***}$ $7.411^{***}$ $7.411^{***}$ <td></td> <td>; 0)</td> <td><math>0.048^{***}</math></td> <td><math>0.046^{***}</math></td> <td><math>0.047^{***}</math></td> <td>-0.000</td> <td>0.002</td> <td>0.002</td> <td><math>0.048^{***}</math></td> <td><math>0.048^{***}</math></td> <td><math>0.049^{***}</math></td>		; 0)	$0.048^{***}$	$0.046^{***}$	$0.047^{***}$	-0.000	0.002	0.002	$0.048^{***}$	$0.048^{***}$	$0.049^{***}$
	0, 0) $0.143 * * $ $0.146 * * $ $-0.025$ $-0.018$ $-0.021$ $0.168 * * $ $0.161 * * * $ $0.1$ 5, 0) $0.127 * * * $ $0.127 * * * $ $0.131 * * * $ $-0.011$ $-0.005$ $-0.004$ $0.138 * * * $ $0.161 * * * $ $0.1$ 1, 0) $0.058 * * * $ $0.061 * * * $ $0.061 * * * $ $-0.011$ $-0.002$ $0.004$ $0.138 * * * $ $0.133 * * * $ $0.133 * * * $ $0.133 * * * $ $0.133 * * * $ $0.133 * * * $ $0.070 * * * $ $0.11 * * * * * * * * * * * * * * * * * * $	_	, 0)	0.022***	0.022***	0.022**	-0.002	-0.001	-0.000	0.024***	0.023***	0.022**
5, 0) $0.127**$ $0.131**$ $0.131**$ $0.133**$ $0.069**$ 0, 0) $6.478**$ $6.932**$ $7.041$ $-0.01$ $-0.00$ $-0.008$ $0.066**$ $0.069**$ $7.411**$ $7.492**$ 5, 0) $9.890**$ $10.526**$ $10.659**$ $2.225**$ $2.362**$ $2.411**$ $4.253**$ $4.570**$ $4.632**$ 5, 0) $0.990**$ $10.549**$ $2.225**$ $2.362**$ $2.411**$ $4.253**$ $4.637**$ $4.637**$ 6, 0) $0.15.492**$ $10.526**$ $10.659**$ $2.922**$ $4.197**$ $4.223**$ $1.637**$ $7.411**$ $7.495**$ 7, 0) $0.132**$ $16.075**$ $16.059**$ $3.922**$ $4.197**$ $4.223**$ $11.570**$ $1.2035**$ 6, 0) $0.132**$ $0.022**$ $0.024**$ $0.025**$ $0.025**$ $0.025**$ $0.025**$ $0.025**$ 1, 0) $0.050**$ $2.73$ $2.73$ $2.73$ $2.73$ $2.73$ $2.73$ $2.73$ 2, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1,	5, 0) $0.127^{***}$ $0.128^{***}$ $0.131^{***}$ $-0.011$ $-0.005$ $-0.004$ $0.138^{***}$ $0.133^{****}$ $0.1$ 1, 0) $0.58^{***}$ $0.060^{***}$ $0.061^{****}$ $-0.011$ $-0.010$ $-0.008$ $0.066^{***}$ $0.070^{****}$ $0.0$ 0.0 0.0 0.0 $0.058^{***}$ $0.060^{***}$ $0.061^{****}$ $-0.011$ $-0.010$ $-0.008$ $0.066^{***}$ $0.70^{****}$ $4.6$ $5, 0)$ $9.890^{***}$ $10.526^{***}$ $10.559^{***}$ $2.352^{***}$ $2.362^{***}$ $2.411^{***}$ $4.257^{***}$ $4.570^{***}$ $12.1$ $1, 0)$ $15.492^{***}$ $16.075^{***}$ $16.558^{***}$ $3.922^{***}$ $4.197^{***}$ $4.223^{***}$ $6.959^{***}$ $11.878^{***}$ $12.1$ $0, 0)$ $0.132^{***}$ $16.075^{***}$ $16.258^{***}$ $3.922^{***}$ $3.164^{***}$ $6.959^{***}$ $11.878^{***}$ $12.1$ $0, 0)$ $0.122^{***}$ $10.570^{***}$ $10.570^{***}$ $11.878^{***}$ $12.1$ $0.002^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $1, 0)$ $0.90^{***}$ $0.025^{***}$ $0.025^{***}$ $0.025^{***}$ $0.025^{***}$ $0.025^{***}$ $0.025^{***}$ $0.0025^{***}$ $0.0025^{***}$ $0.0025^{***}$ $0.0025^{***}$ $0.0025^{***}$ $0.0025^{***}$ $0.00000^{***}$ $0.00000^{***}$ $0.00000^{***}$ $0.00000^{***}$ $0.0000^{***}$ $0.0000^{***}$ $0.0000^{***}$ $0.00000^{***}$ $0.00000^{***}$ $0.00000^{***}$ $0.00000^{***}$ $0.00000^{***}$ $0.000000^{***}$ $0.00000000^{***}$ $0.000000^{***}$ $0.0000000000000000$		0, 0)	0.143***	0.143***	0.146***	-0.025	-0.018	-0.021	0.168***	0.161***	0.177***
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	1, 0) $0.03***$ $0.060***$ $0.061***$ $0.070***$ $0.0$ 0, 0) $6.478***$ $6.932***$ $7.043***$ $2.225***$ $2.362***$ $2.362***$ $4.570***$ $4.6$ 5, 0) $0.980***$ $10.526***$ $10.659***$ $2.225***$ $2.362***$ $4.570***$ $4.6$ 5, 0) $9.800***$ $10.526***$ $10.659***$ $2.225***$ $2.362***$ $4.197***$ $4.253***$ $7.411***$ $7.4$ 1, 0) $15.492***$ $10.659***$ $2.059***$ $2.322***$ $4.197***$ $4.223***$ $11.878***$ $12.1$ 0, 0) $0.132***$ $10.659***$ $2.922***$ $4.197***$ $4.223***$ $11.878***$ $12.1$ 0, 0) $0.132***$ $10.659***$ $2.922***$ $4.197***$ $4.223***$ $12.602***$ 0, 0) $0.132***$ $10.659***$ $0.047***$ $0.073***$ $0.073***$ $0.073***$ 1, 0) $0.030***$ $0.025***$ $0.025***$ $0.025***$ $0.025***$ $0.025***$ 273 $273$ $273$ $273$ $273$ $273$ $273$ 273 $273$ $273$ $273$ $273$ $273$ 273 $273$ $273$ $273$ $273$ $273$ 273 $273$ $273$ $273$ $273$ $273$ 273 $273$ $273$ $273$ $273$ $273$ 273 $273$ $273$ $273$ $273$ $273$ 274 $2.00000****$ $0.025****$ $0.025***$ $0.025***$		; 0)	0.127***	0.128***	0.131***	-0.011	-0.005	-0.004	0.138***	0.133***	0.135***
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	0, 0) $6.478^{***}$ $6.932^{***}$ $7.043^{***}$ $2.225^{***}$ $2.362^{***}$ $2.362^{***}$ $4.570^{***}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{**}$ $7.411^{*}$ $7.411^{*}$ $7.411^{*}$ $7.411^{*}$ $7.411$	_	,0)	0.058***	0.060***	0.061***	-0.011	-0.010	-0.008	0.066***	0.070***	$0.069^{***}$
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	5, 0) $9.800^{***}$ $10.526^{***}$ $10.659^{***}$ $2.931^{***}$ $3.115^{***}$ $3.164^{***}$ $6.959^{***}$ $7.411^{***}$ $7.4$ 1, 0) $15.492^{***}$ $16.075^{***}$ $16.258^{***}$ $3.922^{***}$ $4.197^{***}$ $4.223^{***}$ $11.570^{***}$ $11.878^{***}$ $12.0$ 0, 0) $0.132^{***}$ $16.075^{***}$ $16.258^{***}$ $3.922^{***}$ $4.97^{***}$ $4.223^{***}$ $11.570^{***}$ $11.878^{***}$ $12.0$ 0, 0) $0.132^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.073^{***}$ $0.025^{*}$ $0.025^{*}$ $0.025^{*}$ $0.025^{*}$ $0.025^{*}$ $0.025^{*}$ $0.025^{*}$ $0.025^{*}$ $0.025^{*}$ $0.025^{*}$ $0.025^{*}$ $0.025^{$		0, 0)	6.478***	6.932***	7.043***	2.225***	2.362***	2.411***	4.253***	4.570***	4.632***
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	1,0)       15.492***       16.075***       16.258***       3.922***       4.197***       4.223***       11.570***       11.878***       12.1         0,0)       0.132***       0.055***       0.059***       0.073***       0.073***       12.1         5,0)       0.92***       0.073***       0.073***       0.073***       0.073***       12.1         5,0)       0.92***       0.047***       0.047***       0.075***       0.073***       12.1         1,0)       0.050***       0.025***       0.047***       0.025***       0.045***       0.045***       0.045***       0.045***       13.1       573       273<		; 0)	9.890***	10.526***	10.659***	2.931***	3.115***	3.164***	6.959***	7.411***	7.495***
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	_	,0)	15.492***	16.075***	16.258***	3.922***	4.197***	4.223***	11.570***	11.878***	12.035***
5, 0) 0.022*** 0.047*** 0.047*** 0.045*** 1, 0) 0.050*** 0.025*** 0.025*** 0.025***	5, 0) $0.092^{***}$ $0.045^{***}$ $0.045^{***}$ 1, 0) $0.050^{***}$ $0.045^{***}$ $0.045^{***}$ $1, 0)$ $0.050^{***}$ $0.025^{***}$ $0.025^{***}$ $0.025^{***}$ 273 $273$ $273$ $273$ $273$ $273$ $273$ $273$ $273$ $273$ $273$ $273$ $10, -5, -1)$ days esents the CAARs, AVAR, and CAAV for Schedule 13D, a control sample of Schedule 13G filings, and their difference, $(-10, -5, -1)$ days vert date using the market model (1), the Carhart model (2) and the FF5 model (3). The control sample of Schedule 13G filings is formed by using the watch matching method and matching each event firm to a Schedule 13D event from the same year and the same industry (two-digit SIC) with the bour. The coarsened exact matching is estimated using (log) firm size, market-to-book ratio, return on assets (ROA) measured at $t - 1$ , and the	_	0, 0)	0.132***			0.059***			0.073***		
1, 0) 0.050*** 0.025*** 0.025*** 0.025*** 0.025***	1, 0)       0.050***       0.025***       0.025***         273       273       273       273       273         esents the CAARs, SCAARs, AVAR, and CAAV for Schedule 13D, a control sample of Schedule 13G filings, and their difference, (-10, -5, -1) days vent date using the market model (1), the Carhart model (2) and the FF5 model (3). The control sample of Schedule 13G filings is formed by using the cat matching method and matching each event firm to a Schedule 13D event from the same year and the same industry (two-digit SIC) with the abour. The coarsened exact matching is estimated using (log) firm size, market-to-book ratio, return on assets (ROA) measured at <i>t</i> - 1, and the	I	; 0)	0.092***			$0.047^{***}$			0.045***		
273 273 273 273 273 273	273  273  273  273  273  273  273  273	_	, 0)	$0.050^{***}$			0.025***			0.025***		
	esents the CAARs, SCAARs, AVAR, and CAAV for Schedule 13D, a control sample of Schedule 13G filings, and their difference, $(-10, -5, -1)$ days vent date using the market model (1), the Carhart model (2) and the FF5 model (3). The control sample of Schedule 13G filings is formed by using the vact matching method and matching each event firm to a Schedule 13D event from the same year and the same industry (two-digit SIC) with the hbour. The coarsened exact matching is estimated using (log) firm size, market-to-book ratio, return on assets (ROA) measured at $t - 1$ , and the			273	273	273	273	273	273			
	the target firm ROA measured between years $t = 3$ and $t = 1$ . Panel A presents the results for the full sample. Panel B presents results for the 'clean'	ţħ,	target fin	m ROA measure	ed between years	t-3 and $t-1$ . Pai	nel A presents th	he results for th	e full sample. Pan	el B presents resul	ts for the 'clean	

 Table 13
 Abnormal returns, return volatility and volume of Schedule 13D filings, a control sample of Schedule 13G filings and their difference (continued)

**Figure 7** Total turnover around the main event date: the y axis measures the daily share turnover for the (-30, +30) window around the event date for the full sample of Schedule 13D events (see online version for colours)



**Figure 8** Total turnover around the main event date: the y axis measures the daily share turnover for the (-30, +30) window around the event date for the full sample of Schedule 13G events (see online version for colours)



#### 4.3 Robustness analysis – seasonality effects

In this section, we examine whether our findings may be driven by seasonality. In particular, we examine whether average measures during the 45-trading day disclosure period differ from average measures during the same calendar window in the year prior to the filing date. We therefore use the same methodology and calculate abnormal returns, return volatility and volume for the same time period of the previous year. There are 618 Schedule 13D filings and 930 Schedule 13G filings that have non-missing values for the hypothesised estimation and event period of the previous year. Results are presented in Table 11 and show that both Schedule 13D (Panel A) and 13G filings (Panel B) have

negative abnormal return and volume for the same calendar window in the year prior to the filing date. Schedule 13D filings have a negative (-1.5%) and statistically significant CAAR at the 10% level during the previous year 45-trading day window.

#### 4.4 Robustness analysis – coarsened exact matching for Schedule 13G filings

We turn next to perform another robustness test. We hold the Schedule 13D filings as the main sample and create a control sample of Schedule 13G filings with similar characteristics using the same criteria as in Brav et al. (2018) to do the matching. In particular, the control sample of Schedule 13G filings is formed by using the coarsened exact matching method and matching each event firm to a Schedule 13D event from the same year and the same industry (two-digit SIC) with the nearest neighbour, where the coarsened exact matching is estimated using (log) firm size, market-to-book ratio, return on assets (ROA) measured at t - 1, and the change in the target firm ROA measured between years t - 3 and t - 1. The Market, Carhart and FF5 models are used to calculate abnormal returns. We perform the robustness test for both the filing and the event date. Results are presented in Tables 12 and 13 and our findings are similar to our main findings. In particular, CAARs, SCAARs, AVARs and CAAVs are much higher and statistically significant for both the full sample and 'clean' subsample of Schedule 13D filings when compared to Schedule 13G filings, for all the examined models.

#### 5 Conclusions

In this paper, we exploit a hand-collected dataset of hedge fund interventions to test short-term value creation. The novelty of our approach is that it separates gains arising from insider trading vis-à-vis those generated more broadly by information asymmetry. We examine both Schedule 13D and Schedule 13G filings made by the same hedge funds. We find robust, strong and consistent evidence of positive short-term value creation prior to the filing date and more precisely prior to the event date that the fund surpasses the 5% threshold but only for Schedule 13D filings. In particular, Schedule 13D filings present positive and statistically significant CAARSs, SCAARs and CAAV at the (-45, 0) window from the filing date and the (-10, 0), (-5, 0) and (-1, 0) windows from the event date while AVAR is much greater than one for all the examined windows. Schedule 13D and Schedule 13G filings generate positive abnormal returns in the post filing period which are not statistically different. The latter finding is not surprising since our research design evaluates both 13D and 13G filings on a level playing field. On the other hand, there is no evidence of short-term value creation prior to the filing date for Schedule 13G firms. We find no evidence of abnormal returns for a control sample of similar albeit non-targeted firms across all evaluation windows. Further robustness tests indicate that our findings are not driven by seasonal effects.

The paper's main implication is that hedge funds hold valuable information for their 'active' targets prior to the initial filing even when no plans or proposals are stated which is reflected into the firm's abnormal returns, return volatility and volume. Both 13D and 13G targeted firms present abnormal returns in the post filing period which are significantly lower than the 13D returns prior to the filing date. The implication of this finding is that investors need to exercise caution in moving quickly into a stock

immediately after a 13D filing since most gains appear to have been made before disclosure. Contrary to some evidence in the literature, our findings suggest that there are no significant short-term gains for Schedule 13G filings during the stock building period. Our findings are also consistent with recently proposed changes by the SEC to reduce information asymmetries, and accelerate the filing deadlines for Schedules 13D beneficial ownership reports.

#### References

- Agarwal, V., Jiang, W., Tang, Y. and Yang, B. (2013) 'Uncovering hedge fund skill from the portfolio holdings they hide', *The Journal of Finance*, Vol. 68, No. 2, pp.739–783.
- Ajinkya, B.B. and Jain, P.C. (1989) 'The behavior of daily stock market trading volume', *Journal* of Accounting and Economics, Vol. 11, No. 4, pp.331–359.
- Aragon, G.O., Hertzel, M. and Shi, Z. (2013) 'Why do hedge funds avoid disclosure? Evidence from confidential 13F filings', *Journal of Financial and Quantitative Analysis*, Vol. 48, No. 5, pp.1499–1518.
- Bebchuk, L.A., Jackson Jr, R.J. and Jiang, W. (2013) 'Pre-disclosure accumulations by activist investors: evidence and policy', *The Journal of Corporation Law*, Vol. 39, No. 1, pp.1–34.
- Becht, M., Franks, J., Grant, J. and Wagner, H.F. (2017) 'Returns to hedge fund activism: an international study', *The Review of Financial Studies*, Vol. 30, No. 9, pp.2933–2971.
- Becht, M., Franks, J., Mayer, C. and Rossi, S. (2008) 'Returns to shareholder activism: evidence from a clinical study of the Hermes UK Focus Fund', *The Review of Financial Studies*, Vol. 22, No. 8, pp.3093–3129.
- Ben-David, I., Franzoni, F., Landier, A. and Moussawi, R. (2013) 'Do hedge funds manipulate stock prices?', *The Journal of Finance*, Vol. 68, No. 6, pp.2383–2434.
- Boyson, N.M. and Mooradian, R.M. (2011) 'Corporate governance and hedge fund activism', *Review of Derivatives Research*, Vol. 14, No. 2, pp.169–204.
- Brav, A., Jiang, W., Ma, S. and Tian, X. (2018) 'How does hedge fund activism reshape corporate innovation?', *Journal of Financial Economics*, Vol. 130, No. 2, pp.237–264.
- Brav, A., Jiang, W., Partnoy, F. and Thomas, R. (2008) 'Hedge fund activism, corporate governance, and firm performance', *The Journal of Finance*, Vol. 63, No. 4, pp.1729–1775.
- Brown, S.J. and Warner, J.B. (1980) 'Measuring security price performance', *Journal of Financial Economics*, Vol. 8, No. 3, pp.205–258.
- Brunnermeier, M.K. and Nagel, S. (2004) 'Hedge funds and the technology bubble', *The Journal of Finance*, Vol. 59, No. 5, pp.2013–2040.
- Campbell, C.J. and Wasley, C.E. (1996) 'Measuring abnormal daily trading volume for samples of NYSE/ASE and NASDAQ securities using parametric and nonparametric test statistics', *Review of Quantitative Finance and Accounting*, Vol. 6, No. 3, pp.309–326.
- Clifford, C.P. (2008) 'Value creation or destruction? Hedge funds as shareholder activists', *Journal* of Corporate Finance, Vol. 14, No. 4, pp.323–336.
- Collin-Dufresne, P. and Fos, V. (2015) 'Do prices reveal the presence of informed trading?', *The Journal of Finance*, Vol. 70, No. 4, pp.1555–1582.
- Cready, W.M. and Ramanan, R. (1991) 'The power of tests employing log-transformed volume in detecting abnormal trading', *Journal of Accounting and Economics*, Vol. 14, No. 2, pp.203–214.
- DeFond, M., Hung, M. and Trezevant, R. (2007) 'Investor protection and the information content of annual earnings announcements: international evidence', *Journal of Accounting and Economics*, Vol. 43, No. 1, pp.37–67.

- Devos, E., Hao, W., Prevost, A.K. and Wongchoti, U. (2015) 'Stock return synchronicity and the market response to analyst recommendation revision', *Journal of Banking & Finance*, September, Vol. 58, pp.376–389.
- Dodd, P. and Warner, J. (1983) 'On corporate governance: a study of proxy contests', *Journal of Financial Economics*, Vol. 11, Nos. 1–4, pp.401–438.
- Edmans, A. (2009) 'Blockholders, market efficiency, and managerial myopia', *The Journal of Finance*, Vol. 64, No. 6, pp.2481–2513.
- Gantchev, N. and Jotikasthira, C. (2018) 'Institutional trading and hedge fund activism', *Management Science*, Vol. 64, No. 6, pp.2930–2950.
- Giglia, K. (2016) 'Little letter, a big difference: an empirical inquiry into possible misuse of Schedule 13G/13D filings', *Columbia Law Review*, Vol. 116, No. 1, pp.105–146.
- Gow, I.D., Shin, S.P. and Srinivasan, S. (2014) *Activist Directors: Determinants and Consequences*, Harvard Business School Working Paper, No. 14-120.
- Greenwood, R. and Schor, M. (2009) 'Investor activism and takeovers', *Journal of Financial Economics*, Vol. 92, No. 3, pp.362–375.
- Griffin, J.M. and Xu, J. (2009) 'How smart are the smart guys? A unique view from hedge fund stock holdings', *The Review of Financial Studies*, Vol. 22, No. 7, pp.2531–2570.
- Grossman, S.J. and Stiglitz, J.E. (1980) 'On the impossibility of informationally efficient markets', *The American Economic Review*, Vol. 70, No. 3, pp.393–408.
- Hirshleifer, D., Subrahmanyam, A. and Titman, S. (1994) 'Security analysis and trading patterns when some investors receive information before others', *The Journal of Finance*, Vol. 49, No. 5, pp.1665–1698.
- Ivashina, V. and Sun, Z. (2011) 'Institutional stock trading on loan market information', *Journal of Financial Economics*, Vol. 100, No. 2, pp.284–303.
- Klein, A. and Zur, E. (2009) 'Entrepreneurial shareholder activism: hedge funds and other private investors', *The Journal of Finance*, Vol. 64, No. 1, pp.187–229.
- Landsman, W.R. and Maydew, E.L. (2002) 'Has the information content of quarterly earnings announcements declined in the past three decades?', *Journal of Accounting Research*, Vol. 40, No. 3, pp.797–808.
- Landsman, W.R., Maydew, E.L. and Thornock, J.R. (2012) 'The information content of annual earnings announcements and mandatory adoption of IFRS', *Journal of Accounting and Economics*, Vol. 53, Nos. 1–2, pp.34–54.
- Lipton, M. (2013) Dealing with Activist Hedge Funds, Harvard Law School Forum on Corporate Governance and Financial Regulation Blog [online] https://corpgov.law.harvard.edu /2012/08/09/dealing-with-activist-hedge-funds/ (9 August 2012).
- Massoud, N., Nandy, D., Saunders, A. and Song, K. (2011) 'Do hedge funds trade on private information? Evidence from syndicated lending and short-selling', *Journal of Financial Economics*, Vol. 99, No. 3, pp.477–499.
- Patell, J.M. (1976) 'Corporate forecasts of earnings per share and stock price behavior: empirical test', *Journal of Accounting Research*, Vol. 14, No. 2, pp.246–276.
- Wong, Y.T.F. (2020) 'Wolves at the door: a closer look at hedge fund activism, *Management Science*, Vol. 66, No. 6, pp.2347–2371.

#### Notes

1 We do not suggest that activist hedge funds engage in unlawful insider trading since, tender offers aside, trading on the basis of material; non-public information alone does not violate US law. All we are saying is that activist firms have the potential to exploit material, non-public information around the time they buy a significant block of stock of the target company, and as such their activities may be deemed 'insider trading'.

#### 320 E. Karpouzis et al.

- 2 See https://www.sec.gov/news/press-release/2022-22. As stated by SEC Chair Gary Gensler "These amendments would update our reporting requirements for modern markets, reduce information asymmetries, and address the timeliness of Schedule 13D and 13G filings. Investors currently can withhold market moving information from other shareholders for 10 days after crossing the 5% threshold before filing a Schedule 13D, which creates an information asymmetry between these investors and other shareholders. The filing of Schedule 13D can have a material impact on a company's share price, so it is important that shareholders get that information sooner. The proposed amendments also would clarify when and how certain derivatives acquired with control intent count towards the 5% threshold, clarify group formation, and create related exemptions".
- 3 To the extent that hedge funds are perceived as informed traders who have strong incentives to gather costly information about the target's fundamental value, their intervention may be viewed as enhancing rather than destructing market efficiency, in the sense that prices are more likely to reflect fundamental value rather than short term earnings (see Edmans, 2009).
- 4 General rules and regulations, Securities Exchange Act of 1934, § 240.13d-102.
- 5 The 'reporting persons' are stated in the 'item 2' section ('identity and background') of the Schedule 13D filing.
- 6 Form ADV is the uniform form used by investment advisers to register with both the SEC and state securities authorities.
- 7 Some advisers that are exempt from registration file reports as an exempt reporting adviser. Information provided to ADV forms must be updated periodically.
- 8 The available types of clients are:
  - a individuals (other than high net worth individuals)
  - b high net worth individuals
  - c banking or thrift institutions
  - d investment companies
  - e business development companies
  - f pooled investment vehicles (other than investment companies and business development companies)
  - g pension and profit sharing plans (but not the plan participants or government pension plans)
  - h charitable organisations
  - i state or municipal government entities (including government pension plans)
  - j Other investment advisers
  - k insurance companies
  - 1 sovereign wealth funds and foreign official institutions
  - m corporations or other businesses not listed above
  - n other.
- 9 SIC code 6,000 to 6,999.
- 10 The Williams Act suggests that the filing must be made 'within ten days' after surpassing the 5% threshold. However, it is not stated whether it refers to calendar or trading days. In our analysis, we use the straightforward reading of the Act defining the term 'day' as a one calendar day.
- 11 60 calendar days.
- 12 We would like to thank an anonymous referee for making this suggestion.